

A Service of

ZBW

Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

Neyer, Ulrike; Sterzel, André

Working Paper Capital requirements for government bonds: Implications for bank behaviour and financial stability

DICE Discussion Paper, No. 275

Provided in Cooperation with: Düsseldorf Institute for Competition Economics (DICE), Heinrich Heine University Düsseldorf

Suggested Citation: Neyer, Ulrike; Sterzel, André (2017) : Capital requirements for government bonds: Implications for bank behaviour and financial stability, DICE Discussion Paper, No. 275, ISBN 978-3-86304-274-5, Heinrich Heine University Düsseldorf, Düsseldorf Institute for Competition Economics (DICE), Düsseldorf

This Version is available at: https://hdl.handle.net/10419/171936

Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.

Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



WWW.ECONSTOR.EU

DISCUSSION PAPER

No 275

Capital Requirements for Government Bonds – Implications for Bank Behaviour and Financial Stability

Ulrike Neyer, André Sterzel

December 2017



d|u|p düsseldorf university press

IMPRINT

DICE DISCUSSION PAPER

Published by

düsseldorf university press (dup) on behalf of Heinrich-Heine-Universität Düsseldorf, Faculty of Economics, Düsseldorf Institute for Competition Economics (DICE), Universitätsstraße 1, 40225 Düsseldorf, Germany www.dice.hhu.de

Editor:

Prof. Dr. Hans-Theo Normann Düsseldorf Institute for Competition Economics (DICE) Phone: +49(0) 211-81-15125, e-mail: <u>normann@dice.hhu.de</u>

DICE DISCUSSION PAPER

All rights reserved. Düsseldorf, Germany, 2017

ISSN 2190-9938 (online) - ISBN 978-3-86304-274-5

The working papers published in the Series constitute work in progress circulated to stimulate discussion and critical comments. Views expressed represent exclusively the authors' own opinions and do not necessarily reflect those of the editor.

Capital Requirements for Government Bonds – Implications for Bank Behaviour and Financial Stability

Ulrike Neyer* André Sterzel[†]

December 2017

Abstract

This paper analyses whether the introduction of capital requirements for bank government bond holdings increases financial stability by making the banking sector more resilient to sovereign debt crises. Using a theoretical model, we show that a sudden increase in sovereign default risk may lead to liquidity issues in the banking sector. Our model reveals that in combination with a central bank acting as a lender of last resort, capital requirements for government bonds increase the shock-absorbing capacity of the banking sector and thus the financial stability. The driving force is a regulation-induced change in bank investment behaviour.

JEL classification: G28, G21, G01.

Keywords: bank capital regulation, government bonds, sovereign risk, financial contagion, lender of last resort.

^{*}Heinrich Heine University Düsseldorf Department of Economics, Universitätsstraße 1, 40225 Düsseldorf, Germany, email: ulrike.neyer@hhu.de.

[†]Corresponding author. Heinrich Heine University Düsseldorf, Department of Economics, Universitätsstraße 1, 40225 Düsseldorf, Germany, email: andre.sterzel@hhu.de.

1 Introduction

During the financial crisis of 2007 onwards, significant contagion effects between sovereigns and banks could be observed. Serious doubts about the solvency of some EU member states put pressure on the balance sheets of banks with large sovereign debt exposures. In turn, significant bank bailouts strained public finances. This reinforcing feedback loop led to substantial risks to financial and macroeconomic stability. Against this background, there is an ongoing debate about whether the abolishment of the preferential treatment of sovereign borrowers in banking regulation can mitigate possible contagion effects from sovereigns to banks. Jens Weidmann, the president of the Deutsche Bundesbank, for example, strongly advocates in favour of the abolishment:

"There is one field in regulation, however, where too little has been done so far - the treatment of sovereign exposures in banks' balance sheets. A banking system can only truly be stable if the fate of banks does not hinge on the solvency of their national sovereigns. Thus, I have been advocating, for quite some time now, a phasing-out of the preferential treatment of sovereign borrowers over private debtors." (Weidmann (2016))

Compared to other assets, sovereign debt is given privileged treatment in banking regulation with respect to capital and liquidity requirements as well as to large exposure regimes. This paper deals with the preferential government bond treatment in capital regulation. Although the default probability of some EU member states is significantly higher than zero, banks do not have to back the government bonds of these countries with equity, these bonds are assigned a zero-risk weight in bank capital regulation.¹ Banks have multi-billion euro exposures to sovereign debt, in particular banks in stressed euro area countries have more than doubled their sovereign debt exposures in recent years (see Figure 1). However, banks' sovereign holdings can act as a contagion channel through which sovereign distress can severely affect the banking sector. Considering, for example, the Greek sovereign debt crisis of 2009 onwards, the distressed state of public finances triggered fragility in the banking sector. Against this background the aim of this paper is

¹See CRR (Capital Requirement Regulation) Article 114. The CRR and CRD (Capital Requirement Directive) IV implemented the Basel III accords in EU law.



Figure 1: Banks' sovereign debt holdings as a percentage of their total assets in certain stressed and non-stressed euro area countries. *Data source: ECB.*

to investigate within a theoretical model whether the contagion channel from sovereigns to banks can be weakened through the introduction of capital requirements for government bonds, thereby making the banking sector more resilient to sovereign debt crises.

In a first step, we analyse the banks' investment and financing behaviour in different capital regulation scenarios. The banks' objective is to maximise their depositors' expected utility. The depositors have the usual Diamond-Dybvig preferences which imply that there is no aggregate liquidity risk in the banking sector, though banks do face idiosyncratic liquidity risks. Banks can invest in a risk-free short-term asset, which earns no return, and in two risky long-term assets (government bonds and loans) with an expected positive return. However, whereas loans are totally illiquid, government bonds are highly liquid as there exists an interbank market for this asset. Investing in government bonds thus allows banks to hedge their idiosyncratic liquidity risks.² Besides deposits, banks can raise equity capital to finance their investments. Raising costly equity capital allows banks to transfer liquidity risks associated with highly profitable but totally illiquid loans from risk-averse depositors to risk-neutral investors, thus increasing their depositors' expected utility. Within this model framework, we first investigate the impact of capital

²As pointed out by Gennaioli et al. (2014a), for example, banks may hold government bonds for many different reasons. So government bonds do play an important role in managing a banks' daily activities. In this context our model thus captures the idea that banks hold government bonds to manage their liquidity.

requirements only for loans on the bank investment and financing behaviour. It is shown that the introduction of this capital regulation induces banks not only to raise more equity but also to increase their loan investment. The reason is that the regulation-induced reduction of the loan-to-equity ratio implies that the potential for a beneficial liquidity risk transfer from depositors to investors is no longer fully exploited. Banks mitigate the resulting negative effect on depositors' expected utility by increasing their investment in highly profitable loans in absolute terms. However, this means that banks must raise more equity to fulfil the capital requirements. Consequently, introducing capital requirements for loans implies that the loan-to-equity ratio decreases, but in absolute terms both loans and equity increase. This bank behaviour (increasing loan investment and raising more equity capital) will be reinforced if government bonds also have to be backed with equity. The crucial point is that this additional capital requirement implies that the loan-to-equity ratio will decrease even further so that the potential for a beneficial liquidity risk transfer can be exploited even less.

In a second step, we then investigate the banks' shock-absorbing capacity in different capital regulation scenarios. We suppose that the economy is hit by a shock in the form of an increase in the default probability of sovereign bonds (government bond shock). The increased doubts about sovereign solvency may lead to a sovereign bond price drop and hence to liquidity issues in the banking sector, leading to illiquid but per se solvent banks going bankrupt. We show that capital requirements themselves cannot prevent illiquid but solvent banks from going bankrupt. However, combined with a central bank as a lender of last resort (LOLR) that provides additional liquidity against adequate collateral, the introduction of capital requirements for loans and additional capital requirements for government bonds increase the banking sector's shock-absorbing capacity. The driving force is that the regulation-induced change in bank investment behaviour (more loans) implies that banks have more adequate collateral from which to obtain additional liquidity from the LOLR.

The rest of the paper is structured as follows. Section 2 presents the related literature. Section 3 describes the model setup. Section 4 analyses both sides of the interbank market for government bonds and derives the market equilibrium. Section 5 outlines the liquidity-risk-transfer property of equity capital and derives the banks' optimal investment and financing behaviour in different capital regulation scenarios. Building on these analyses, Section 6 discusses the consequences of capital requirements for the shock-absorbing capacity of the banking sector and the importance of the central bank acting as a LOLR in this context. The final section summarises the paper.

2 Related Literature

Our paper contributes to three strands of literature. The first strand deals with financial contagion, the second with different institutions aiming to weaken the financial contagion channel between sovereigns and banks, and the third with the influence of capital requirements on bank behaviour.

In the literature, there is no single definition of financial contagion. We will refer to financial contagion if financial linkages imply that a shock, which initially affects only one or a few firms (financial or non-financial), one region or one sector of an economy, spreads to other firms, regions or sectors. In a seminal paper, Allen and Gale (2000) show that if there is an interbank deposit market which allows banks to balance their different liquidity needs, a small liquidity preference shock initially affecting only one bank may spread to other banks, leading to the breakdown of the whole banking sector. Allen and Carletti (2006) model contagion effects from the insurance to the banking sector. The crucial point is that the credit-risk transfer between these sectors implies that banks and insurance companies hold the same securities. A crisis in the insurance sector forces the insurance companies to sell these securities. The resulting price drop of these assets then also affects banks' balance sheets leading to severe problems in the banking sector. In a similar vein Heyde and Neyer (2010) show that credit-risk transfer within the banking sector may create a channel of financial contagion. Allen and Gale (2006) extend the Allen-Carletti model, enabling them to analyse the impact of bank capital regulation on systemic risk. They show that the introduction of binding capital requirements may increase systemic risk, as they induce inefficiencies for banks. These inefficiencies can be mitigated if banks share risks with the insurance sector. However, this risk-sharing increases contagion potentials between banks and insurance companies. Especially since the financial crisis of 2007 onwards, there has

been a growing literature on financial contagion between sovereigns and banks.³ Gennaioli et al. (2014b) identify banks' government bond holdings as a potential link through which a sovereign default can severely affect the banking sector. However, they claim that it is exactly the existence of this potential contagion channel which makes the occurrence of a sovereign default less likely (and thus the occurrence of a banking crisis triggered by respective contagion). They argue that banks, especially those in countries with highly developed financial institutions, hold large amounts of domestic government bonds. This means that governments do not have an incentive to strategically default because, against conventional wisdom, it is not possible to discriminate between foreign and domestic bond holders, so that a sovereign default would badly hit the domestic banking sector and thus the domestic economy. In other words, such a default would be too costly.⁴ In a similar way, Bolton and Jeanne (2011) analyse how much of an impact a sovereign default has on the banking sector in financially integrated economies. They find that, on the one hand, financial integration leads to risk diversification benefits for banks. However, on the other hand it generates a financial contagion channel between sovereigns. Acharya et al. (2014) investigate the two-way-feedback risk transmission between sovereigns and banks. They argue that government bank bailouts lead to a rise in sovereign credit risk. This in turn weakens the banking sector as the value of the banks' sovereign bond holdings and the value of their implicit and explicit government guarantees decrease. Cooper and Nikolov (2013) also examine the diabolic loop between sovereigns and banks and, in this context, the role played by fiscal guarantees and equity capital. As a policy implication the authors stress the role of equity capital as an important regulatory tool to isolate banks from sovereign risk and they suggest the implementation of capital requirements on sovereign exposures. Broner et al. (2014) argue that in turbulent times, sovereign debt offers a higher expected return to domestic than to foreign creditors. This creditor discrimination implies that banks increase their investment in domestic government bonds. If banks are financially constrained, which is especially the case in turbulent times, this

 $^{^{3}}$ For a survey of the main channels through which sovereign risk influences the banking sector, see European Systemic Risk Board (2015).

⁴Acharya and Rajan (2013) analyse why countries have an incentive to serve their debt even if a government default would lead to little direct domestic cost. They argue that through a default, governments would lose access to debt markets, which would result in a decrease in fiscal spending and therefore in GDP, so that even short-horizon governments have an incentive to repay their debt.

bank investment behaviour will be in line with a decrease of private sector loans (crowdingout effect). This bank investment behaviour not only reduces economic growth but also reinforces the risk of financial contagion between sovereigns and domestic banks.

There is also a rapidly growing empirical literature on financial contagion between banks and sovereigns. Acharya and Steffen (2015), for example, provide empirical evidence, that especially large, low-capitalised banks with risky assets tend to invest in longterm, risky, peripheral⁵ government bonds, financing these investments by borrowing from the unsecured short-term wholesale market ("carry trade" behaviour). This financing and investment behaviour can be motivated by banks' risk-shifting incentives and regulatory capital arbitrage as sovereign exposures are not subject to capital regulation. In a similar vein, Korte and Steffen (2015) empirically show that within the euro area, the zero-risk weight on (risky) sovereign debt exposures in banking regulation reinforces a potential financial contagion channel from foreign sovereigns to domestic banks, as the preferential treatment of government bonds in bank capital regulation subsidises risky sovereign debt holdings, so that banks hold a relatively large amount of these assets.⁶ A huge number of empirical analyses confirms a strong link between the potential default of sovereigns and of banks by deriving a strong positive correlation between the CDS spreads of sovereigns and banks.⁷

Based on the literature on the sovereign-bank doom loop, the second strand of related literature discusses different newly implemented or proposed institutions aiming to weaken the financial contagion channel between sovereigns and banks. Covi and Eydam (2016) argue that the *new recovery and resolution framework* actually weakens this contagion channel as, due to a "bail-in" rule, bank insolvencies no longer strain public finances. Farhi and Tirole (2016) claim that national governments may favour a lax banking supervision, particularly in times of a weak domestic banking sector, as losses in the banking industry can be shifted to international investors. They thus argue that a *shared supranational*

⁵The term "peripheral" refers to the euro area (Greece, Italy, Ireland, Portugal, Spain, or GIIPS).

⁶Note in this context, that also Haubrich and Wachtel (1993) find empirical evidence that the introduction of a risk-adjusted capital ratio with a zero-risk weight on sovereign exposures makes sovereign bonds more attractive for banks in relation to loans. Accordingly, high-indebted banks in particular increase their sovereign bond investment at the expense of loans in order to meet the required capital ratio.

⁷See, for example, Acharya et al. (2014), Altavilla et al. (2017), Alter and Beyer (2014), Angelini et al. (2014), De Bruyckere et al. (2013), Ejsing and Lemke (2011), Gennaioli et al. (2014a), Gross and Kok (2013) or Vergote (2016).

banking supervision can diminish contagion effects between internationally operating banks and sovereigns. Brunnermeier et al. (2016) develop a model which illustrates how to isolate banks from sovereign risk via the introduction of *European Safe Bonds* ("ESBies") issued by a European debt agency. They argue that holding these bonds disentangles banks from potential sovereign defaults, as ESBies are backed by a well-diversified portfolio of euro-area government bonds and are additionally senior on repayments.

The third strand of related literature deals with the influence of capital requirements on bank behaviour. Blum (1999), for example, points out that a binding required riskweighted capital ratio may increase banks' risk taking behaviour.⁸ Hyun and Rhee (2011) find that the introduction of a binding required risk-adjusted capital ratio may imply that banks reduce their loan supply (instead of increasing their equity capital) to fulfil the capital requirement. Harris et al. (2014) use a general equilibrium model to discuss the welfare consequences of higher bank capital requirements. They find that higher capital requirements may incentivise banks to invest in high-risk projects, inducing a decrease in overall welfare.

Our paper contributes to all three strands of literature: In our theoretical analysis, banks hold highly liquid government bonds to hedge their idiosyncratic liquidity risks. These government bond holdings generate a potential for financial contagion from sovereigns to banks (first strand). With respect to institutions aiming to weaken the financial contagion channel between banks and sovereigns (second strand), we derive that the introduction of binding capital requirements in general, and for government bond holdings in particular, are not sufficient to weaken this channel, but in addition, a central bank acting as a LOLR is necessary. Furthermore, we show how capital requirements influence bank investment and financing behaviour (third strand). The introduction of binding capital requirements only for loans implies that the banks' loan-to-equity ratio (obviously) decreases, but in absolute terms both loans and equity increase. This bank behaviour is reinforced if government bonds also become subject to capital requirements.

⁸Flannery (1989) and Furlong and Keeley (1989) also analyse the risk-taking incentives associated with higher capital requirements. However, their analysis focuses on the role played by the existence of a deposit insurance.

3 Model

3.1 Technology

We consider three dates, t = 0, 1, 2 and a single all-purpose good that can be invested or consumed. At date 0, the all-purpose good can be invested in three types of assets: one short-term and two long-term assets. The short-term asset represents a simple storage technology i.e. one unit at date 0 returns one unit at date 1. The two long-term assets are government bonds and loans. However, unlike in other theoretical works, government bonds are not completely safe but yield a random return S. With probability $1 - \beta$ the investment fails and one unit of the all-purpose good invested in government bonds at date 0 produces only l < 1 units of this good at date 2. With probability β , the investment succeeds and produces h > 1 units at date 2. A government bond is a liquid asset and can be traded at price p on an interbank market at date 1. The loan portfolio yields a random return K. If the loan investment succeeds, one unit invested at date 0 will generate a return of H > h > 1 units at date 2 with probability $\alpha < \beta$. With probability $(1 - \alpha)$ the investment fails and produces only L < l < 1 units at date 2. The main characteristics of the loan portfolio are that it is the asset with the highest expected return as E(K) > E(S) > 1, the highest risk as the variance Var(K) > Var(S), and that it is totally illiquid as loans cannot be traded at date 1. Banks discover whether the long-term assets succeed or fail at date 2. Table 1 summarises the returns on the different types of assets.

	Return at date 1	Return at date 2
Short-term asset	1	
Government bonds	p	$\begin{array}{ccc} h & & \beta \\ & & \end{array} \right\} E(S) > 1$
		$l (1-\beta) \int \frac{1}{2} \left(1-\beta\right) dt$
Loan portfolio	0	$H \qquad \alpha \qquad \Big\} \qquad E(K) > E(S) Var(K) > Var(S)$
		$L (1-\alpha) \int L(K) > L(S), \forall u \in K \} > \forall u \in S,$

Table 1: Return on the Different Types of Assets (Investment at Date 0: 1 Unit)

3.2 Agents and Preferences

In our model, there are three types of agents: a continuum of risk-averse consumers normalised to one, a large number of banks and a large number of risk-neutral investors. Each consumer is endowed with one unit of the all-purpose good at date 0. Like in Diamond and Dybvig (1983) consumers can be categorised into two groups. One group values consumption only at date 1 (early consumers), the other group only at date 2 (late consumers). We assume both groups are the same size. The proportion of early consumers is $\gamma = 0.5$ and the proportion of late consumers is $(1 - \gamma) = 0.5$. Denoting a consumer's consumption by c, his utility of consumption is described by

$$U(c) = ln(c). \tag{1}$$

However, at date 0 each consumer is unsure of their liquidity preference. He does not know whether he is an early or late consumer. Therefore, he concludes a deposit contract with a bank. According to this contract, he will deposit his one unit of the all-purpose good with the bank at date 0 and can withdraw c_1^* units of the all-purpose good at date 1 or c_2^* units of this good at date 2. As we have a competitive banking sector, each bank invests in the short-term asset and the two long-term assets in a way that maximises its depositors' expected utility.

While there is no aggregate liquidity risk (the fraction of early consumers is $\gamma = 0.5$ for sure) banks are subject to idiosyncratic liquidity risk. Accordingly, they do not know their individual proportion of early consumers. With probability ω a bank has a fraction γ_1 of early consumers and with probability $(1 - \omega)$ a bank faces a fraction γ_2 ($\gamma_2 > \gamma_1$) of early consumers, so that $\gamma = 0.5 = \omega \gamma_1 + (1 - \omega) \gamma_2$. As in Allen and Carletti (2006), we assume the extreme case in which $\gamma_1 = 0$ and $\gamma_2 = 1$, so that $\omega = 0.5$.⁹ Because of this strong assumption, we have two types of banks; banks with only early consumers (*early banks*) and banks with only late consumers (*late banks*) and the probability of becoming an early or a late bank is 0.5 each. Banks can hedge their idiosyncratic liquidity risk by

⁹The reason for this strong assumption is to keep the optimisation problem as simple as possible. Without this assumption the expected utility function given by (3) would be: $E(U) = \omega \gamma_1 ln(c_1) + (1 - \omega)\gamma_2 ln(c_1) + \omega(1 - \gamma_1)[\alpha\beta ln(c_{2Hh}) + \alpha(1 - \beta)ln(c_{2Hl}) + (1 - \alpha)\beta ln(c_{2Lh}) + (1 - \alpha)(1 - \beta)ln(c_{2Ll})] + (1 - \omega)(1 - \gamma_2)[\alpha\beta ln(c_{2Hh}) + \alpha(1 - \beta)ln(c_{2Hl}) + (1 - \alpha)\beta ln(c_{2Lh}) + (1 - \alpha)(1 - \beta)ln(c_{2Ll})].$ Given $\gamma_1 = 0$ and $\gamma_2 = 1$ the first and the last term of the equation can be eliminated.

using an interbank market for government bonds: All banks invest in government bonds and the short-term asset at date 0. At date 1, when each bank has learnt whether it is an early or a late bank, it sells or buys government bonds in exchange for the short-term asset on the interbank market to balance its individual liquidity position.

In addition to the funds from consumers, banks have the opportunity to raise funds (equity capital) from risk-neutral investors. These investors are endowed with an unbounded amount of capital W_0 at date 0. The contract concluded between a bank and an investor defines the units of the all-purpose good (equity capital) which are provided at date 0 ($e_0^* \ge 0$) and the units which are repaid and consumed by the investor at date 1 and date 2 ($e_1^* \ge 0$ and $e_2^* \ge 0$). As in Allen and Carletti (2006) the utility function of a risk-neutral investor is given by

$$U(e_0, e_1, e_2) = \rho(W_0 - e_0) + e_1 + e_2, \tag{2}$$

where the parameter ρ presents the investor's opportunity costs of investing in the banking sector.

3.3 Optimisation Problem

As ex-ante, i.e. at date 0, all banks are identical, we can consider a representative bank when analysing the banks' optimal investment and financing behaviour at date 0. Deposits are exogenous and equal to one. The bank has to decide on units x to be invested in the short-term asset, on units y to be invested in government bonds, on units u to be invested in loans and on units e_0 to be raised from the risk-neutral investors. A bank's optimal behaviour requires the maximisation of the expected utility of its risk-averse depositors. Consequently, a bank's optimisation problem reads

$$\max E(U) = 0.5ln(c_1) + 0.5[\alpha\beta ln(c_{2Hh}) + \alpha(1-\beta)ln(c_{2Hl}) + (1-\alpha)\beta ln(c_{2Lh}) + (1-\alpha)(1-\beta)ln(c_{2Ll})]$$
(3)

with
$$c_1 = x + yp$$
, (4)

$$c_{2Hh} = uH + \left(\frac{x}{p} + y\right)h - e_{2Hh},\tag{5}$$

$$c_{2Hl} = uH + \left(\frac{x}{p} + y\right)l - e_{2Hl},\tag{6}$$

$$c_{2Lh} = uL + \left(\frac{x}{p} + y\right)h - e_{2Lh},\tag{7}$$

$$c_{2Ll} = uL + \left(\frac{x}{p} + y\right)l - e_{2Ll},\tag{8}$$

s.t.
$$\rho e_0 = 0.5(\alpha e_{2H} + (1 - \alpha)e_{2L}) + 0.5(\alpha \beta e_{2Hh})$$

$$+ \alpha (1 - \beta) e_{2Hl} + (1 - \alpha) \beta e_{2Lh} + (1 - \alpha) (1 - \beta) e_{2Ll}), \tag{9}$$

$$CR^{min} = \frac{e_0}{\phi_x x + \phi_y y + \phi_u u},\tag{10}$$

$$e_0 + 1 = x + y + u, \tag{11}$$

$$x, y, u, e_0, e_{2Hh}, e_{2Hl}, e_{2Lh}, e_{2Ll} \ge 0.$$
(12)

Equation (3) describes the expected utility of the bank's depositors. With probability 0.5 the bank is an early bank, i.e. all of its depositors are early consumers who thus withdraw their deposits at date 1. In this case, the bank will use the proceeds of the short-term asset $(x \cdot 1)$ and of selling all its government bonds on the interbank market $(y \cdot p)$ to satisfy its depositors, as formally revealed by (4).

With probability 0.5, the bank is a late bank, i.e. all of its depositors are late consumers and withdraw their deposits at date 2. The consumption level of a late consumer depends on the returns on the bank's investments in government bonds and loans. As the probabilities of the success of these investments, α and β , are independent, we can identify four possible states: both investments succeed, only the investment in the loan portfolio succeeds, only the investment in the government bonds succeeds, or both investments fail. We denote these four states simply as Hh, Hl, Lh Ll. Equations (5) to (8) represent the consumption levels of late depositors in these possible states. The first term on the righthand side in each of these equations shows the proceeds from the investment in loans, the second from the investment in government bonds. Note that the quantity of government bonds a late bank holds at date 2 consists of the units y it invested itself in government bonds at date 0, and of those it has bought on the interbank market in exchange for its units of the short-term asset x at date 1. The last term depicts the amount a bank has to pay to the risk-neutral investors at date 2. Due to their risk-neutrality, they are indifferent between whether to consume at date 1 or at date 2. Consequently, optimal (risk-averse) consumer contracts require $e_1^* = 0$.

Equation (9) represents the investors' incentive-compatibility constraint. Investors are only willing to provide equity capital e_0 to the banking sector if at least their opportunity costs ρ are covered. With probability 0.5 the bank is an early bank. Then, it will use its total amount of x including those units obtained in exchange for its total amount of government bonds on the interbank market to satisfy all its depositors at date 1, while investors will receive the total proceeds from loans $e_{2H} = uH$ or $e_{2L} = uL$ at date 2. With probability 0.5, the bank is a late bank. Then, the bank will buy government bonds on the interbank market in exchange for its short-term assets at date 1. At date 2, it will repay its depositors and investors. The investors will receive a residual payment from the proceeds of the bank's total loan and government bond investment, i.e. those returns not being used to satisfy the bank's depositors. Constraint (10) captures the capital requirements the bank may face. They are expressed as a minimum capital ratio CR^{min} of the bank's equity e_0 to its (risk-)weighted assets $\phi_x x + \phi_y y + \phi_u u$. If $\phi_x = \phi_y = \phi_u = 0$, there will be no capital requirements. If $\phi_x = \phi_y = 0$ and $\phi_u > 0$, there will be a privileged treatment of (risky) government bonds as only loans are subject to financial regulation. This privileged treatment will be repealed if $\phi_y > 0$. Then, risky government bonds will also have to be backed with equity capital. The budget constraint is represented in equation (11), and the last constraint (12) represents the non-negativity constraint.

4 Interbank Market for Government Bonds

Before solving the banks' optimisation problem in the next section, we will have a closer look at the interbank market for government bonds. Banks use government bonds to balance their idiosyncratic liquidity needs: At date 0 all banks invest in government bonds and at date 1 the early banks sell their government bonds to the late banks in exchange for the short-term asset. We assume that the late consumers' expected utility of an investment in risky government bonds is higher than that of an investment in the safe short-term asset i.e.

$$\beta ln(h) + (1 - \beta) ln(l) \ge ln(1) = 0.$$
(13)

This means that the expected return on government bonds is sufficiently higher than on the short-term asset to compensate the risk-averse depositors for the higher risk. If it were not for this assumption, an interbank market for government bonds would not exist as no bank would invest in government bonds.

At date 1, each bank has learnt whether it is an early bank or a late bank. However, late banks will only buy government bonds in exchange for their short-term asset if

$$\beta ln(h) + (1 - \beta)ln(l) - ln(p) \ge ln(1), \tag{14}$$

i.e. if the expected utility of their depositors is than at least as high as that from the alternative of storing the short-term asset until date 2. Consequently, there is a maximum price

$$p^{max} = h^{\beta} l^{(1-\beta)} \tag{15}$$

late banks are willing to pay for a government bond. If $p \leq p^{max}$, a late bank wants to sell the total amount of its short-term asset in exchange for government bonds as government bonds yield a (weakly) higher expected utility for their depositors. If $p > p^{max}$, a late bank does not want to sell any unit of its short-term assets in exchange for government bonds.

Note that at date 0, all banks are identical and solve the same optimisation problem. Accordingly, for all banks the optimal quantities invested in the short-term asset and the long-term assets are identical. We denote these optimal quantities by x^* , y^* , and u^* . Considering the number of depositors is normalised to one, the optimal quantities of each individual bank correspond to the respective aggregate quantities invested in each asset type. As half of the banks are late banks, aggregate demand for government bonds at date 1 is

$$y^{D} = \begin{cases} 0.5 \frac{x^{*}}{p} & \text{if } p \leq p^{max}, \\ 0 & \text{if } p > p^{max}. \end{cases}$$
(16)

Figure 2 illustrates this demand function. The jump discontinuity at p^{max} results from the fact that for $p \leq p^{max}$ late banks want to sell their total amount of the short-term asset $0.5x^*$ in exchange for government bonds. The demand curve is downward sloping because the amount of liquidity in the banking sector which can be used for buying government bonds is limited to $0.5x^*$. Consequently, a higher price p implies that fewer government bonds can be bought. Independently of the price, early banks want to sell all their government bonds at date 1 as early consumers only value consumption at this time. Therefore, the aggregate supply of government bonds is perfectly price inelastic. The respective aggregate supply curve is given by

$$y^S = 0.5y^*$$
 (17)

as illustrated graphically in Figure 2.

Considering (16) and (17) and denoting the equilibrium price for government bonds p^{**} ,¹⁰ the market clearing condition becomes

$$\frac{x^*}{p^{**}} = y^*.$$
 (18)

As there is no aggregate liquidity uncertainty and as all banks solve the same optimisation problem at date 0, aggregate supply and demand and thus the equilibrium variables are known at date 0. In addition, the following considerations reveal that $p^{**} = 1$. If $p^{**} < 1$, the return on government bonds at date 1 would be negative and thus smaller than on the short-term asset. Consequently, at date 0 banks would invest only in the short-term asset and not in government bonds. However, if no bank buys government bonds at date 0,

¹⁰To be able to distinguish between those quantities optimally invested in the different assets and those quantities exchanged in equilibrium on the interbank market, we index optimal variables with * and equilibrium variables with **.

there will be no supply of government bonds and thus no interbank market for government bonds with a positive price at date 1.

If $p^{**} > 1$, a government bond would be worth more than the short-term asset at date 1. Therefore, no bank would invest in the short-term asset at date 0 but only in government bonds. However, if at date 0 no bank invests in the short-term asset but only in government bonds, there will be no demand for government bonds at date 1, and thus no interbank market for this asset with a positive price. Consequently, the only possible equilibrium price at date 1 is $p^{**} = 1$. Note that due to (13) and (15), $p^{max} \ge 1$, which implies that the interbank market is always cleared.



Figure 2: Interbank Market for Government Bonds at Date 1

Considering aggregate demand and supply curves allow us to determine the surplus of the banking sector from interbank trading. The equilibrium government bond trading volume is denoted by y^{**} . The blue area reflects the surplus of the late banks. They benefit from interbank trading as the exchange of the short-term asset for government bonds leads to a higher expected utility of their depositors (see equation (13)). The green area shows the surplus of the early banks from interbank trading. At date 1, government bonds produce no return so that their exchange in short-term assets allows for a higher date-1 consumption and thus a higher utility of early depositors.

5 Optimal Bank Investment and Financing Behaviour

This section analyses the impact of repealing the preferential treatment of government bonds in bank capital regulation on bank investment and financing behaviour. However, as a starting point we analyse bank behaviour with and without the possibility of raising equity capital. This allows us to point to the key property of equity capital in our model, which is the property to transfer liquidity risk from risk-averse depositors to risk-neutral investors and thereby to increase depositors' expected utility. The subsequent analysis shows that the introduction of binding capital requirements for loans implies that the potential for liquidity risk transfer is no longer fully exploited which lowers depositors' expected utility. This effect will be reinforced if government bonds also have to be backed with equity. To compensate at least partially for this negative effect, banks invest more in highly profitable loans. To meet the required capital ratio, this changed investment behaviour works alongside raising more equity. Consequently, introducing capital requirements leads to a lower loan-to-equity ratio, but in absolute terms loan investment and equity capital increase.

To demonstrate a bank's optimal investment and financing behaviour in different regulation scenarios, we make use of a numerical example similar to the one used by Allen and Carletti (2006). The government bond returns h = 1.3 with probability $\beta = 0.98$ and l = 0.3 with probability $(1-\beta) = 0.02$. Consequently, the investment in government bonds of one unit of the consumption good at date 0 yields the expected return E(S) = 1.2746 at date 2. Loans are also state-dependent and return at date 2. They return H = 1.54 with probability $\alpha = 0.93$, and they fail and yield L = 0.25 with probability $(1 - \alpha) = 0.07$. Hence, the expected loan return at date 2 is E(K) = 1.4497. Investors' opportunity costs are $\rho = 1.5$.

5.1 No Equity Capital

In the case with no equity capital, the constraints (9) and (10) are omitted, all $e_{(\cdot)} = 0$, and the budget restriction (11) becomes x + y + u = 1. Optimal banking behaviour in this case is shown in Table 2.

Balance Sheet

Α					L
x^*	0.5	50%	D	1	100%
y^*	0.5	50%			
\sum	1	100%	\sum	1	100%

Deposit Contracts:

$$c_1^* = 1$$
 $c_{2h}^* = 1.3$ $c_{2l}^* = 0.3$
 $E(U) = 0.1165$

Proof. See Proof I in Appendix A

Table 2: No Equity Capital: Banks' Optimal Balance Sheet Structure and Repayments to Depositors

Without having the opportunity to raise equity capital, banks invest their total amount of deposits in the short-term asset and in government bonds i.e., only in liquid assets. They do not grant loans.¹¹ Basically, investing in loans has two effects on consumers consumption: First, it increases expected consumption at date 2 as the expected loan return is higher than that of government bonds. Second, it decreases consumption at date 1 as, due to the budget constraint (11), an increase in loan investment implies a respective decrease of investment in liquid assets, and early consumers are only repaid with the proceeds of the liquid assets. In our numerical example, the effects of the loans' illiquidity on consumption is so strong that even at point u = 0 the marginal utility from date-1 consumption exceeds the expected marginal utility from date-2 consumption, i.e. the non-negativity constraint on u becomes binding.

Moreover, banks divide their investment equally into the liquid assets, $x^* = y^*$. With respect to this result two aspects are important. First, one half of the banks are early banks whereas the other half are late banks. Second, there is idiosyncratic but no aggregate liquidity uncertainty, so that at date 0 banks are certain of the aggregate supply and demand in the government bond market and therefore know the equilibrium price

¹¹Diamond and Dybvig (1983) consider the explicit role of banks in an economy in the sense that banks transform illiquid assets into liquid liabilities i.e. banks allow better risk-sharing among consumers with different consumption preferences. However, in our numerical example without equity capital, banks do not invest in illiquid loans i.e. banks are obsolete in this case. However, with the introduction of equity capital, banks start to invest in illiquid assets and create liquidity, i.e. there is a role for banks in our subsequent analyses.

 $p^{**} = 1$ (see Section 4 for details). Accordingly, all banks invest the identical amount in government bonds and in the short-term asset, to be able to hedge their idiosyncratic liquidity risks completely by trading government bonds on the interbank market at date 1 when consumption uncertainty is resolved. This allows us to set $x^* = y^* = 0.5z^*$ in our subsequent analyses. The variable z^* then donates a bank's optimal investment in liquid assets (short-term asset and government bonds).

5.2 With Equity Capital

If banks have the opportunity to raise equity capital from investors, but do not face a binding minimum capital ratio ($CR^{min} = 0$), we will get the solutions given in Table 3 for optimal bank behaviour. The results show that even if the banking sector is not

Balance Sheet					
Α					L
x^*	0.4544	41.87%	e_0^*	0.0853	7.86%
y^*	0.4544	41.87%	D	1	92.14%
u^*	0.1765	16.26%			
\sum	1.0853	100%	\sum	1.0853	100%

Contracts with Investors:

early banks:
$$e_{2H}^* = 0.2718$$
 $e_{2L}^* = 0.0441$ late banks: $e_{2Hh}^* = 0$ $e_{2Hl}^* = 0$ $e_{2Lh}^* = 0$

Deposit Contracts:

$$c_1^*{=}\;0.9088\quad c_{2Hh}^*{=}1.4532\quad c_{2Hl}^*{=}\;0.5444\quad c_{2Lh}^*{=}\;1.2256\quad c_{2Ll}^*{=}0.3168$$

$$E(U){=}0.1230$$

Proof. See Proof II in Appendix A

Table 3: With Equity Capital: Banks' Optimal Balance Sheet Structure and Repayments to Investors and Depositors

subject to capital requirements, it will be optimal for banks to raise equity capital. Since $\rho > E(K)$, equity capital is costly for banks in the sense that investors' opportunity costs, and thus the amount banks expect to repay to investors, exceed the expected return even of the banks' most profitable asset – in our case, loans. Nevertheless, banks have an

incentive to raise equity capital as it allows the liquidity risk involved with an investment in relatively high profitable loans to be transferred from risk-averse depositors to riskneutral investors, leading to an increase in depositors' expected utility $(E(U))_{no \ capital} <$ $E(U)|_{with \ capital, CR^{min}=0}$, see tables 2 and 3).

A crucial point is that with the possibility to raise equity, the banks' budget constraint is softened. In the case without this possibility (Section 5.1), an increase in loan investment leads to a decrease of investment in liquid assets to the same amount, $\left(\frac{\partial z}{\partial u}\Big|_{no \ capital} = -1\right)$, and thus to a respective decline in date-1 consumption. However, with the possibility of raising equity capital, an increase in loans leads to a lower necessary decrease in liquid assets, $\frac{\partial z}{\partial u}|_{with \ capital, CR^{min}=0} > -1.^{12}$ Consequently, an investment in relatively highly profitable but illiquid loans, which increases expected date-2 consumption, only implies a relatively small decrease of consumption at date 1, so that there is an increase in depositors' expected utility.¹³ It is crucial that a huge part of the additional loan investment is financed by raising equity capital from risk-neutral investors. Due to their risk-neutrality it is optimal that they bear the liquidity risk involved with the banks' loan investment. The risk-averse depositors' thus benefit from a liquidity risk transfer to the risk-neutral investors.

Optimal risk-sharing implies that if it turns out that a bank is an early bank, the investors of this bank will receive the total proceeds from the loan investment at date 2 $(e_{2H}^*, e_{2L}^* > 0)$. If it turns out that a bank is a late bank, they will receive nothing $(e_{2Hh}^*, e_{2Hl}^*, e_{2Lh}^*, e_{2Ll}^* = 0)$. Considering investors thus get repaid with the total proceeds from the bank loan investment but only with probability 0.5, and that their opportunity costs are higher than the expected return on loans $(\rho > E(K))$, the bank loan investment must exceed the amount of raised equity capital to be able to satisfy investors' claims. This means that it is not possible to finance an additional loan investment exclusively by raising more equity, i.e. an increase in loan investment is still associated with a decrease of investment in liquid assets $(-1 < \frac{\partial z}{\partial u}|_{with \ capital, CR^{min}=0} < 0)$. Formally, the

¹²In our numerical example $\frac{\partial z}{\partial u}|_{with \ capital, CR^{min}=0} = -0.5168 > -1.$ ¹³Formally, with the possibility of raising equity capital, at point u = 0 the expected marginal utility from date-2 consumption exceeds the marginal utility from date-1 consumption, and it is optimal for banks to increase their investment in loans relative to their investment in liquid assets. This bank behaviour increases depositors' expected date-2 consumption and decreases their date-1 consumption and thus balances the marginal utilities.

investors' incentive-compatibility constraint given by (9) becomes $e_0^* \rho = 0.5u^* E(K)$, so that $\frac{2\rho}{E(K)} = \frac{u^*}{e_0^*}|_{CR^{min}=0}$. This means that the loan investment needs to be at least $\frac{2\rho}{E(K)}$ times higher than the amount of raised equity capital. The expression shows that the optimal investment in loans relative to the raised equity capital must be higher the lower the expected returns are of the loans compared to the investors' opportunity costs.¹⁴ In our numerical example $\frac{2\rho}{E(K)} = \frac{u^*}{e_0^*}|_{CR^{min}=0} = 2.0692.$

5.3 Binding Capital Requirement for Loans

In this section, we analyse bank behaviour when banks face a required minimum capital ratio with a preferential treatment of risky sovereign bonds in the sense that only risky loans have to be backed with equity. We introduce a risk weight of 1 for this asset class, i.e. in constraint (10) we have $\phi_x = \phi_y = 0$ and $\phi_u = 1$. In our analysis, we suppose a binding required minimum capital ratio $CR^{min}|_{\phi_u=1,\phi_x=\phi_y=0} = \frac{e_0}{u} = 0.5799.^{15}$ The results for optimal bank behaviour under this additional constraint are shown in Table 4. The comparison of the results for optimal bank behaviour given in tables 3 and 4 reveals that the capital requirement induces banks to raise more equity and to increase their investments in loans and liquid assets (e_0^* , u^* , and z^* increase). In the following, we will have a closer look at this bank reaction. We divide our analysis of the banks' adjustment behaviour into two steps. In the first step, we look at the direct consequences of the introduced capital regulation for bank investment and financing behaviour and thus for depositors' consumption possibilities. In a second step, we then analyse the banks' optimal response to these changed consumption possibilities.

First step: In Section 5.2 it has been shown that without capital requirements an investor will only get repaid if it turns out that his bank is an early bank. Then, at date 2, he receives the total proceeds from his bank's loan investment. However, the capital

¹⁴In our model, the probability of becoming an early bank is $(1 - \omega = 0.5)$, see Section 3. Not inserting 0.5 for $1 - \omega$, we have $\frac{\rho}{(1-\omega)E(K)} = \frac{u^*}{c_0^*}|_{CR^{min}=0}$ which reveals that investment in loans relative to the raised equity capital must also be the higher the lower the probability of becoming an early bank is, i.e. the lower the probability is that the investor actually gets repaid.

¹⁵If banks do not face binding capital requirements (Section 5.2) they choose an optimal capital ratio of: $CR^{opt} = \frac{c_0^*}{u^*} = \frac{0.0853}{0.1765} = 0.4833$. In order to analyse the impact of a binding capital ratio, $CR^{min} > CR^{opt}$ must hold. We consider a binding minimum capital ratio which is 20% higher than CR^{opt} .

Balance Sheet					
Α					L
$x^{*} =$	0.4581	41.06%	$e_0^* =$	0.1157	10.37%
$y^{*} =$	0.4581	41.06%	D =	1	89.63%
$u^* =$	0.1995	17.88%			
\sum	1.1157	100%	\sum	1.1157	100%

Contracts with Investors:

 $\begin{array}{c|c} \mbox{early banks:} & e_{2H}^* = 0.3072 & e_{2L}^* = 0.0499 \\ \hline \mbox{late banks:} & e_{2Hh}^* = 0.0635 & e_{2Hl}^* = 0 & e_{2Lh}^* = 0 & e_{2Ll}^* = 0 \end{array}$

Deposit Contracts:

 $c_1^* = 0.9162$ $c_{2Hh}^* = 1.4348$ $c_{2Hl}^* = 0.5821$ $c_{2Lh}^* = 1.2409$ $c_{2Ll}^* = 0.3247$ E(U) = 0.1224

Proof. See Proof III in Appendix A

Table 4: Binding Capital Requirement for Loans: Banks' Optimal Balance Sheet Structure and Repayments to Investors and Depositors

regulation-induced decrease of $\frac{u}{e_0}$ implies that this is no longer possible.¹⁶ The decreased loan-to-equity ratio implies that the expected returns from loans are no longer sufficient to satisfy investors' claims. In Section 5.2 we have shown that this repayment scheme requires an optimal loan investment to be at least $\frac{2\rho}{E(K)} = \frac{u^*}{e_0^*}|_{CR^{min}=0}$ times higher than the raised equity capital. However, $\frac{u^*}{e_0^*}|_{CR^{min}|_{\phi_u=1,\phi_x=\phi_y=0}} < \frac{u^*}{e_0^*}|_{CR^{min}=0}$.¹⁷ Consequently, with the capital requirement not only early banks but also late banks have to pay a positive amount to their investors $(e_{2Hh}^*|_{CR^{min}=0} = 0$ but $e_{2Hh}^*|_{CR^{min}|_{\phi_u=1,\phi_x=\phi_y=0}} > 0)$. This reduces depositors' expected date-2 consumption as they have to share the returns from the longterm assets with the investors. Moreover, the budget constraint implies that the required decrease in $\frac{u}{e_0}$ leads to an increase in bank investment in liquid assets (z increases). This increases depositors' date-1 consumption. Accordingly, in the first step of our analysis the

 $[\]boxed{\begin{array}{c} {}^{16}\text{Note that the required loan-to-equity ratio is the reciprocal of the minimum capital ratio} \\ {\left(\frac{u}{e_0}\right|_{CR^{min}|\phi_u=1,\phi_x=\phi_y=0}=\frac{1}{CR^{min}|\phi_u=1,\phi_x=\phi_y=0}\right)}.$ ${}^{17}\text{In our numerical example } \frac{u^*}{e_0^*}|_{CR^{min}|\phi_u=1,\phi_x=\phi_y=0}=1.7243<\frac{u^*}{e_0^*}|_{CR^{min}=0}=2.0692.$

introduced capital requirement for loans implies that the expected marginal utility from date-2 consumption exceeds the marginal utility from date-1 consumption.

Second step: To remove this inefficiency, banks increase loans and equity according to the required minimum loan-to-equity ratio $\frac{u}{e_0}|_{CR^{min}|_{\phi_u=1,\phi_x=\phi_y=0}}$. As this ratio is larger than one (in our numerical example it is equal to 1.7243), the balance sheet constraint implies that investment in liquid assets decreases again. This second-step bank behaviour leads to a higher (lower) date-2 consumption (date-1 consumption) until the marginal utility from date-1 consumption again equals the expected marginal utility from date-2 consumption. Since the increase in liquid assets in the first step exceeds the decrease in the second step, in absolute terms there is an overall increase in liquid assets.

Consequently, the introduction of the binding required minimum capital ratio implies that the potential for liquidity risk transfer is not fully exploited, $\frac{u^*}{e_0^*}|_{CR^{min}|_{\phi_u=1,\phi_x=\phi_y=0}} < \frac{u^*}{e_0^*}|_{CR^{min}=0}$, and that the banks' loan-to-liquid asset ratio increases, $\frac{u^*}{z^*}|_{CR^{min}|_{\phi_u=1,\phi_x=\phi_y=0}} > \frac{u^*}{z^*}|_{CR^{min}=0}$. The restricted possibility of transferring liquidity risk from depositors to investors leads to a lower depositors' expected utility, $E(U)|_{CR^{min}=0} > E(U)|_{CR^{min}|_{\phi_u=1,\phi_x=\phi_y=0}}$.

5.4 Binding Capital Requirements for Loans and Government Bonds

This section analyses bank optimal investment and financing behaviour when the required minimum capital ratio also includes a positive risk weight for government bond holdings, i.e. when risky sovereign exposures are also subject to capital regulation. We assume a risk weight for government bonds of $\phi_y = 0.05$.¹⁸ The risk weights for loans and the short-term asset are the same as in Section 5.3, i.e. $\phi_u = 1$ and $\phi_x = 0$. Also the level of the required minimum capital ratio is not changed, $CR^{min} = 0.5799$. Hence, the capital regulation constraint becomes $CR^{min}|_{\phi_u=1,\phi_y=0.05,\phi_x=0} = \frac{e_0}{u+0.05y} = 0.5799$. The resulting optimal bank behaviour is illustrated in Table 5.

Comparing the results given in tables 4 and 5 shows that repealing the preferential treatment of government bonds in capital regulation induces banks to raise additional equity capital, to grant more loans and to increase their liquid asset holdings (e_0^* , u^* , and z^* increase). Consequently, the bank reaction is the same as when a binding capital ratio

¹⁸By assuming that $\phi_u = 1 > \phi_y = 0.05$, we consider sovereign bonds to be less risky than loans.

Balance Sheet					
Α					L
$x^{*} =$	0.4586	40.02%	$e_0^* =$	0.1459	12.73%
$y^{*} =$	0.4586	40.02%	D =	1	87.27%
$u^* =$	0.2287	19.96%			
\sum	1.1459	100%	\sum	1.1459	100%

Contracts with Investors:

 $\begin{array}{c|c} \mbox{early banks:} & e_{2H}^* = 0.3522 & e_{2L}^* = 0.0572 \\ \hline \mbox{late banks:} & e_{2Hh}^* = 0.1166 & e_{2Hl}^* = 0 & e_{2Lh}^* = 0 & e_{2Ll}^* = 0 \\ \end{array}$

Deposit Contracts:

 $c_1^* = 0.9172$ $c_{2Hh}^* = 1.4281$ $c_{2Hl}^* = 0.6274$ $c_{2Lh}^* = 1.2496$ $c_{2Ll}^* = 0.3323$ E(U) = 0.1217

Proof. See Proof IV in Appendix A

Table 5: Capital Requirements for Loans and Government Bonds: Banks' Optimal Balance Sheet Structure and Repayments to Investors and Depositors

is introduced only for loans, as described in Section 5.3. Also, the reason behind this reaction is the same: if banks also have to back government bonds with equity, they can invest less in highly profitable loans per units of costly equity capital, $\frac{u^*}{e_0^*}$ decreases.¹⁹

Accordingly, as in the case in which only loans have to be backed with equity, the expected returns from loans are not sufficient to satisfy investors' claims, so that not only early but also late banks have to make a positive payment to the investors. Also, government bonds have to be backed with equity, this payment must be even higher because of the lower loan-to-equity ratio $\frac{u^*}{e_0^*}$, i.e. $e_{2Hh}^*|_{CR^{min}=0} = 0 < e_{2Hh}^*|_{CR^{min}|_{\phi_u=1,\phi_x=\phi_y=0}} < e_{2Hh}^*|_{CR^{min}|_{\phi_u=1,\phi_y=0.05,\phi_x=0}}$. Consequently, with respect to the optimal bank investment and financing behaviour the same adjustment process as described in detail in Section 5.3 applies. This means that the potential to transfer liquidity risk from depositors to investors will be used even less, $\frac{u^*}{e_0^*}|_{CR^{min}|_{\phi_u=1,\phi_y=0.05,\phi_x=0}} < \frac{u^*}{e_0^*}|_{CR^{min}|_{\phi_u=1,\phi_x=\phi_y=0}} < \frac{u^*}{e_0^*}|_{CR^{min}|_{\phi_u=1,\phi_x=\phi_y=0}} < \frac{u^*}{e_0^*}|_{CR^{min}|_{\phi_u=1,\phi_x=\phi_y=0}} < \frac{u^*}{e_0^*}|_{CR^{min}|_{\phi_u=1,\phi_y=0.05,\phi_x=0}} < \frac{u^*}{e_0^*}|_{CR^{min}|_{\phi_u=1,\phi_x=\phi_y=0}} < \frac{u^*}{e_0^*}|_{CR^{min}|_{\phi_u=1,\phi_y=0.05,\phi_x=0}} < \frac{u^*}{e_0^*}|_{CR^{min}|_{\phi_u=1,\phi_x=\phi_y=0}} < \frac{u^*}{e_0^*}|_{CR^{min}=0}$

 $[\]frac{1^{9} \text{Given that } CR^{min} = \frac{e_{0}}{u+0.05y}, \text{ the positive risk weight for sovereign bonds implies for } y^{*} > 0 \text{ that}}{\frac{u^{*}}{e_{0}^{*}}|_{CR^{min}|_{\phi_{u}=1,\phi_{y}=0.05,\phi_{x}=0}} < \frac{u^{*}}{e_{0}^{*}}|_{CR^{min}|_{\phi_{u}=1,\phi_{x}=\phi_{y}=0}}. \text{ In our numerical example } \frac{u^{*}}{e_{0}^{*}}|_{CR^{min}=0} = 2.0692 > \frac{u^{*}}{e_{0}^{*}}|_{CR^{min}|_{\phi_{u}=1,\phi_{x}=\phi_{y}=0}} = 1.7243 > \frac{u^{*}}{e_{0}^{*}}|_{CR^{min}|_{\phi_{u}=1,\phi_{y}=0.05,\phi_{x}=0}} = 1.5675.$

and the banks' loan-to-liquid asset ratio increases further, $\frac{u^*}{z^*}|_{CR^{min}|_{\phi_u=1,\phi_y=0.05,\phi_x=0}} > \frac{u^*}{z^*}|_{CR^{min}|_{\phi_u=1,\phi_x=\phi_y=0}} > \frac{u^*}{z^*}|_{CR^{min}=0}$. The stronger restriction of transferring liquidity risk from depositors to investors further reduces the depositors' expected utility, $E(U)|_{CR^{min}=0} > E(U)|_{CR^{min}|_{\phi_u=1,\phi_x=\phi_y=0}} > E(U)|_{CR^{min}|_{\phi_u=1,\phi_y=0.05,\phi_x=0}}.$

6 Financial Stability

The aim of this paper is to analyse the resilience of the banking sector in case of a sovereign debt crisis under different capital regulation scenarios. This section shows that increasing doubts about sovereign solvency may lead to liquidity issues in the banking sector driven by a respective price drop for sovereign bonds. A central bank acting as a LOLR can avoid bank insolvencies due to liquidity issues. It turns out that in the presence of a LOLR the abolishment of the preferential treatment of sovereign bonds in financial regulation strengthens the resilience of the banking sector in the case of a sovereign debt crisis.

6.1 Government Bond Shock

After the banks have made their financing and investment decisions at date 0, but before the start of interbank trading at date 1, the economy is hit by a shock in the form of an increase in the default probability of government bonds (we refer to this shock as a government bond shock). This implies a respective decrease of the expected return on government bonds. Denoting after-shock variables with a bar, we thus have $(1 - \overline{\beta}) >$ $(1-\beta)$ and $E(S) > \overline{E(S)}$. As the liquidity shock in Allen and Gale (2000), this government bond shock is assigned a zero probability at date 0, when investment decisions are made. The expected return on the loan portfolio and the return on the short-term asset are not affected by the shock.²⁰

The shock influences the late banks' demand for sovereign bonds in the interbank market at date 1. The decline in the expected return on government bonds implies that the maximum price late banks are willing to pay for a bond decreases (equations (15)

²⁰To keep the model as simple as possible, we assume that the expected loan return is not affected by the government bond shock. However, there is empirical evidence that there are spillovers going from sovereigns to other sectors of an economy (see e.g. Corsetti et al. (2013)) as sovereigns' ratings normally apply as a "sovereign floor" for the ratings assigned to private borrowers. Nevertheless, if we take this correlation into account our results will not qualitatively change. See footnote 27 for details.

and (16)). The early banks' supply of government bonds is not affected by the shock. As their depositors only value consumption at date 1, they want to sell their total holdings of government bonds at the same time, independent of their default probability (see equation (17)).

To be able to satisfy their depositors according to the deposit contract, the price an early bank receives for a government bond must be at least one, i.e. we have a critical price

$$p^{crit} = 1. (19)$$

Setting in equation (15) p^{max} equal to p^{crit} and then solving the equation for $(1 - \beta)$ gives us the critical default probability

$$(1-\beta)^{crit} = \frac{\ln(h) - \ln(p^{crit})}{\ln(h) - \ln(l)} = \frac{\ln(h)}{\ln(h) - \ln(l)}.$$
(20)

If the after-shock default probability of government bonds exceeds this critical probability, the expected return on government bonds will become so low that the maximum price late banks are willing to pay for a bond will fall below one, early banks will be illiquid and insolvent. Therefore, the threshold $(1-\beta)^{crit}$ allows us to distinguish between a small and a large government shock. A small government shock implies that $(1-\overline{\beta^{small}}) \leq (1-\beta)^{crit}$, whereas a large shock means that $(1-\overline{\beta^{large}}) > (1-\beta)^{crit}$. In the following, we will comment on the consequences of both shocks in more detail.

Small Government Bond Shock

Figure 3 illustrates the interbank market for a small government bond shock. The increased sovereign default probability induces that the maximum price late banks are willing to pay for sovereign bonds decreases, $\overline{p^{max \ small}} < p^{max}$. However, the equilibrium price and the equilibrium transaction volume after a small shock do not change, $\overline{p^{**small}} = p^{**} = 1$, $\overline{y^{**small}} = y^{**} = 0.5y^*$. Consequently, the small shock only implies a decline in the late banks' surplus from interbank trading as at the same price late banks receive the same quantity of government bonds but they yield a lower expected return. This is illustrated

by the blue shaded area in Figure 3. The early banks' surplus does not change as their depositors value consumption only at date 1 so that for them the decreased expected (date-2) return on government bonds plays no role.



Figure 3: Interbank Market for Government Bonds at Date 1; $(1 - \overline{\beta^{small}}) \leq (1 - \beta)^{crit}$

In the following, we will discuss in more detail who actually bears the costs of a small government bond shock. Early-bank depositors are not affected by the shock as there is neither a shock-induced change in the equilibrium price, nor in the equilibrium transaction volume on the interbank market for government bonds, so that their consumption does not change:

$$\overline{c_1^{small}} = x^* + y^* \overline{p^{**small}} = x^* + y^* p^{**} = x^* + y^* = c_1^*.$$
(21)

Early-bank investors are not affected by the shock either as they are only repaid from the proceeds of the loan portfolio. However, the shock influences late-bank depositors as due to the decreased expected return on government bonds their expected date-2 consumption decreases.²¹ Whether late-bank investors are affected depends on whether there is a binding capital requirement. If there is no binding capital requirement, the shock will not impact late-bank investors as then, independent of the shock, their repayment will anyhow be equal to zero (see Section 5.2). However, binding capital requirements imply that if both, government bonds and the loan portfolio succeed (state Hh), also late-bank investors will get some repayment at date 2 (see Section 5.3 and 5.4). As the shock implies that the occurrence probability of this event becomes smaller, their expected date-2 repayment decreases.

Large Government Bond Shock

Figure 4 illustrates the interbank market for a large government bond shock. The increase in the government bonds' default probability is so high that their expected return becomes low enough that the maximum price late banks are willing to pay for a bond falls below one, so that considering equation (15), the after-shock equilibrium price becomes

$$\overline{p^{**large}} = \overline{p^{max\ large}} < 1.$$

At the equilibrium price $\overline{p^{**large}}$, there is an excess demand for government bonds²² but the equilibrium trading volume has not changed, $\overline{y^{**large}} = y^{**} = 0.5y^*$. As $\overline{p^{max\ large}} = \overline{p^{**large}}$ the late-banks' surplus from interbank trading becomes zero. In addition, the fall of the equilibrium price ($\overline{p^{**large}} < p^{**}$) also leads to a decrease of the early banks' surplus from interbank trading. They receive a smaller quantity of the short-term asset in exchange for their total holdings of government bonds. The decrease of the equilibrium price below 1 means that early banks are no longer able to fulfil their deposit contracts:

$$\overline{c_1^{large}} = x^* + y^* \overline{p^{**large}} < x^* + y^* p^{**} = x^* + y^* = c_1^*.$$
(23)

²¹Note, that nevertheless late banks do not become insolvent as they can still fulfil the contracts with their depositors as the contractually agreed repayments are not influenced by the shock, $\overline{c_{2(\cdot)}^{small}} = c_{2(\cdot)}^*$ (see equations (5)-(8)).

²²The reason is that late banks want to sell their total holdings of the short-term asset $(0.5x^*)$ in exchange for government bonds. However, the supply of government bonds is limited to the early banks' total holdings of this asset $(0.5y^*)$, so that at prices below 1, there is an excess demand.



Figure 4: Interbank Market for Government Bonds at Date 1; $(1 - \overline{\beta^{large}}) > (1 - \beta)^{crit}$

Early banks are thus insolvent and are liquidated at date 1. In contrast to the small government bond shock, depositors and investors of both early and late banks are affected by the large shock. Early-bank depositors suffer as their date-1 consumption decreases $(\overline{c_1^{large}} < c_1^*)$ and early-bank investors suffer as the loan portfolio's liquidation value at date 1 is zero, so that early-bank investors get no repayment at all.²³ With respect to the late-bank depositors and investors the same argument as in the small-shock scenario holds. Depositors are affected by the shock as their expected date-2 consumption decreases due to the decreased expected return on government bonds. Investors will be affected by the shock if there is a binding capital ratio, as only then may they be repaid, but the probability of actually being repaid declines.

 $^{^{23}}$ If one assumes that a liquidator keeps the loan portfolio until date 2, so that a positive return on the portfolio is realised (either *H* of *L*), it does not prevent the early banks from going bankrupt at date 1. However, the early-bank investors would not be affected by the shock. Note that it is not possible to let the investors bear the costs of the shock instead of the depositors by paying them the proceeds of the loan portfolio at date 2 as they only value consumption at date 1.

6.2 Central Bank as a Lender of Last Resort

As banks hold government bonds in order to hedge their idiosyncratic liquidity risks, a government bond price drop may lead to liquidity issues for banks and thus to insolvencies. To avoid bankruptcies due to liquidity issues we introduce a central bank as a lender of last resort (LOLR) in the sense of Bagehot (1873).²⁴ The central bank provides liquidity to troubled banks against adequate collateral. In our model, banks' loan portfolios serve as collateral. In order to avoid any potential losses for the central bank, the maximum amount of liquidity ψ the central bank is willing to provide to an early bank against its loan portfolio as collateral is²⁵

$$\psi = u^* L. \tag{24}$$

An early bank's additional liquidity needs after a large government bond shock τ are determined by the repayment agreed upon in the deposit contract c_1^* and the lower aftershock repayment $\overline{c_1^{large}}$ (without a LOLR):

$$\tau = c_1^* - \overline{c_1^{large}} = y^* (p^{**} - \overline{p^{**large}}) = y^* (1 - \overline{p^{**large}}).$$
(25)

Equation (25) reveals that the bank's additional liquidity needs increase in its holdings of government bonds y^* and in the extent of the shock which is reflected by the decrease of the government bond price $(p^{**} - \overline{p^{**large}})$. The promised repayment to early consumers c_1^* increases in a bank's holdings of government bonds $(c_1^* = x^* + y^*p^{**} = x^* + y^*)$. The shock-induced price drop for government bonds below one therefore implies the additional liquidity needs the larger the higher the bank's holdings of government bonds y^* are.

 $^{^{24}}$ Bagehot (1873): In a liquidity crisis, a central bank should lend freely, at a high rate of interest relative to the pre-crisis period, to any borrower with good collateral.

²⁵Considering that potential interest payments for the additional central bank liquidity should also be covered by collateral, does not qualitatively change our results. In that case, the maximum amount of liquidity ψ the central bank is willing to provide against the loan portfolio as collateral decreases. This decrease implies that the shock-absorbing capacity of the banking sector in the presence of a LOLR (SAC^{LOLR}) becomes smaller in both regulation scenarios i.e. with and without a preferential regulatory treatment of government bonds. However, as the loan-to-liquid assets ratio $\frac{u^*}{\tilde{t}^*}$ is higher without a preferential treatment of sovereign debt in bank capital regulation, the $SAC^{LOL\tilde{R}}$ will be higher in this case (see equation (30)).

The comparison of a bank's additional liquidity needs τ with the maximum amount of liquidity the central bank is willing to provide ψ gives us the critical government bond price

$$p^{critLOLR} = 1 - \frac{u^*L}{y^*} < 1.$$
 (26)

Inserting $p^{critLOLR}$ for $\overline{p^{**large}}$ in equation (22) and then solving the equation for $(1 - \overline{\beta^{large}})$, gives us the critical default probability

$$(1-\beta)^{critLOLR} = \frac{\ln(h) - \ln(p^{critLOLR})}{\ln(h) - \ln(l)} = \frac{\ln(h) + \ln(\frac{u^*L}{y^*})}{\ln(h) - \ln(l)} = \frac{\ln(h) + \ln(\frac{u^*}{z^*}2L)}{\ln(h) - \ln(l)}.$$
 (27)

If the government bond shock is so large that $(1 - \overline{\beta^{large}}) > (1 - \beta)^{critLOLR}$, the equilibrium price $\overline{p^{**large}}$ will fall below $p^{critLOLR}$, and early banks will become insolvent, despite the existence of a LOLR. The reason is that the central bank is only willing to provide liquidity to illiquid but not to insolvent banks.²⁶ The liquidity issue leads to a solvency issue as the price drop, and thus the resulting early banks' liquidity problem, will be so huge that they will not have sufficient collateral to obtain enough liquidity from the LOLR.

Comparing the critical default probability with and without a central bank as a LOLR (see equations (20) and (27)) reveals the obvious result that with a LOLR the critical default probability is higher. However, the comparison also shows that with a LOLR the critical default probability not only depends on the possible government bond returns h and l, as is the case without a LOLR, but, in addition, on the loan portfolio return L and the bank's investment in government bonds y^* and loans u^* . An increase in y^* leads to a decrease of the critical default probability as then an early bank needs more additional liquidity after a government bond shock (see equation (25)). The critical default probability increases in u^* and L, as then an early bank's collateral increases in quantity and value, so that in case of a shock it can obtain more additional liquidity from the central

²⁶Even if one assumes that the central bank cannot distinguish between illiquid and insolvent banks, the main results do not change. Providing liquidity to insolvent banks does not prevent their insolvency as the maximum liquidity the central bank is willing to provide will not be sufficient to cover the additional liquidity needs of insolvent early banks ($\tau > \psi$).

bank (see equation (24)).²⁷ This has important implications for the banking sector's shock-absorbing capacity under the different capital regulation approaches as we will see in Section 6.3.

In the following we discuss who will actually bear the costs of a government bond shock if the central bank acts as a lender of last resort and if $(1 - \beta)^{crit} < (1 - \overline{\beta}^{large}) \le$ $(1 - \beta)^{critLOLR}$. Early-bank depositors are not affected by the shock, their consumption does not change:

$$\overline{c_1^{large}} = x^* + y^* \overline{p^{**large}} + \tau = x^* + y^* = c_1^*$$
(28)

Early-bank investors, however, are affected by the shock as they get repaid from the proceeds of the loan portfolio and a part of these proceeds has to be used to repay the central bank. With respect to the late-bank depositors and investors the same arguments as in the scenarios without a LOLR hold. Depositors are affected by the shock as their expected date-2 consumption decreases. Investors will be affected if there is a binding capital requirement.²⁸ However, if $(1 - \overline{\beta^{large}}) > (1 - \beta)^{critLOLR}$ early banks are insolvent and thus the central bank does not provide liquidity. Then depositors and investors of both, early and late banks are affected by the shock and the identical arguments hold as in the large-shock scenario without a LOLR.

²⁷We argued at the beginning of this section that considering a possible spillover of the government bond shock to loans would not lead to a qualitative change of our results. In case the probability of loan success is negatively affected by the government bond shock, i.e. if $\alpha > \overline{\alpha}$, the expected consumers' consumption at date 2 will decrease. However, there are no liquidity issues for late banks as the contractually agreed repayments to the consumers are not influenced. The crucial point is that the potential increase in α neither induces a change in the liquidity provision by the central bank (ψ) nor does it lead to an additional liquidity demand (τ). As these variables determine the shock-absorbing capacity with a LOLR (see Section 6.3), spillover effects from sovereign to loans have no impact on our results.

²⁸Note that again late-bank depositors are *only* affected by the shock due to the decreased expected return on government bonds. Late banks do not borrow any additional liquidity from the central bank so that they do not have to use part of the proceeds from the loan portfolio to repay the central bank.

6.3 The Shock Absorbing Capacity of the Banking Sector in Different Capital Regulation Scenarios

The above analysis allows us to discuss the (government bond) shock-absorbing capacity of the banking sector, and in this sense its stability,²⁹ in two different capital regulation scenarios. The difference between the critical and the initial default probability of government bonds serves as a measure of the banking sector's shock-absorbing capacity. The measure thus shows how large a government bond shock can be without implying the insolvency of early banks and thus of a huge part of the banking sector. Considering equations (20) and (27) and denoting the shock-absorbing capacity by SAC and SAC^{LOLR} respectively, we get

$$SAC = (1 - \beta)^{crit} - (1 - \beta) = \frac{\ln(h)}{\ln(h) - \ln(l)} - (1 - \beta)$$
(29)

for the banking sector's shock absorbing capacity without a LOLR and

$$SAC^{LOLR} = (1 - \beta)^{critLOLR} - (1 - \beta) = \frac{\ln(h) + \ln(\frac{u^*}{z^*}2L))}{\ln(h) - \ln(l)} - (1 - \beta)$$
(30)

for the banking sector's shock absorbing capacity with a LOLR.

Equation (29) reveals that without a LOLR, the shock-absorbing capacity is not at all influenced by capital requirements. The reason is that without a LOLR early banks will become insolvent if the maximum price for government bonds that late banks are willing to pay, and thus also the equilibrium price for government bonds, falls below one. Then, early banks will no longer be able to satisfy their customers according to the deposit contract. However, the maximum price late banks are willing to pay is only determined by the expected return on a government bond (see equation (15)) which will not change if capital requirements are introduced. Hence, if there is no LOLR, the shock-induced liquidity problem cannot be solved by any kind of capital requirements, the difference $(1 - \beta)^{crit} - (1 - \beta) = SAC$ is always the same. This result is illustrated in Figure 5 by the solid line.

²⁹The ECB defines financial stability as a condition in which the financial system – intermediaries, markets and market infrastructures – can withstand shocks without major distribution in financial intermediation and the general supply of financial services.

However, as shown by (30), with a LOLR binding capital requirements influence the banking sector's shock-absorbing capacity. The reason is that binding capital requirements influence bank investment behaviour. They imply an increase in both government bond investments y^* and loan investments u^* . The former implies an increase in the banks' additional liquidity needs τ after the shock (see equation (25)), and thus lowers the shock-absorbing capacity. The latter leads to an increase in the additional liquidity ψ the central bank is willing to provide, and therefore also in the shock absorbing capacity. However, the increase in u^* is stronger than in y^* (and thus than in z^* , see Section 5) which implies that there is an overall increase in the SAC^{LOLR} . The main driver for this result is that the regulation-induced increase in u^* implies that banks have more collateral to obtain additional liquidity from the central bank after a respective shock. As the increase in $\frac{u^*}{z^*}$ will be higher if not only loans but also sovereign bonds have to be backed with equity $\left(\frac{u^*}{z^*}|_{CR^{min}|_{\phi_u=1,\phi_y=0.05,\phi_x=0}} > \frac{u^*}{z^*}|_{CR^{min}|_{\phi_u=1,\phi_x=\phi_y=0}} > \frac{u^*}{z^*}|_{CR^{min}=0}\right)$ the (government bond) shock-absorbing capacity of the banking sector will be the highest if there is a LOLR and if loans and sovereign bonds have to be backed with equity $(SAC^{LOLR}|_{CR^{min}|_{\phi_u=1,\phi_y=0.05,\phi_x=0}} > SAC^{LOLR}|_{CR^{min}|_{\phi_u=1,\phi_x=\phi_y=0}} > SAC^{LOLR}|_{CR^{min}=0}).$ This result is illustrated in Figure 5 by the broken line.



Figure 5: Government Bond Shock-Absorbing Capacity of the Banking Sector

7 Summary

In many countries within the EU, banks hold large undiversified amounts of government bonds in their portfolios. These bank sovereign exposures can act as a significant financial contagion channel between sovereigns and banks. The European sovereign debt crisis of 2010 onwards highlighted that some countries within the EU were having severe problems with repaying or refinancing their debt. The resulting price drops of sovereign bonds severely strained the banks' balance sheets. Against this background, there is an ongoing debate about whether the abolishment of the preferential treatment of government bonds in bank capital regulation (sovereign debt from EU member states are still considered as risk-free and thus do not have to be backed with equity capital) can weaken this financial contagion channel between sovereigns and banks. Our paper adds to this debate in two ways. First, by analysing the consequences of introducing capital requirements for sovereign bonds, for bank investment and financing behaviour. Second, by investigating how much these additional capital requirements thus contribute to making the banking sector more resilient against sovereign debt crises.

As pointed out, for example, by Gennaioli et al. (2014a) an important reason for relatively large government bond holdings is that banks use them to manage their everyday business. Capturing this idea, in the model presented in this paper, banks hold government bonds to balance their idiosyncratic liquidity needs by using an interbank market for government bonds. Increasing sovereign solvency doubts may induce a price drop for government bonds, implying liquidity issues in the banking sector leading to bank insolvencies as illiquid banks have no opportunity to obtain additional liquidity. Also, in this paper government bond holdings thus create a financial contagion channel. Our model shows that in the absence of a LOLR the introduction of capital requirements in general and for government bonds in particular are not able to weaken this financial contagion channel. However, this will be the case, i.e. the introduction of capital requirements for government bonds will thus make the sector more resilient to sovereign debt crisis, if there is a LOLR. The reason is that banks can obtain additional liquidity from the LOLR against adequate collateral. Per se illiquid loans serve as adequate collateral, and the introduction of capital requirements for government bonds induces banks to increase their investment in these loans. This means that they will be able to get more additional liquidity in case of financial contagion from the sovereign.

Our model shows that on the one hand the introduction of capital requirements also for government bonds leads to a decrease of depositors' expected utility as binding capital requirements restrict the possibility of a beneficial liquidity risk transfer from depositors to investors. However, on the other hand these additional capital requirements will contribute to a more resilient banking sector in case of a sovereign debt crisis conditioned on the existence of a LOLR. In this context, it should be noted that our paper does not allow for a comprehensive welfare analysis of introducing capital requirements for sovereign bond – and it was also not the aim of the paper.

A Appendix A

Proof I. Using the Lagrangian \mathcal{L} , the bank's optimisation problem can be formulated as

$$\max_{x,y,u} \mathcal{L} = 0.5ln(c_1) + 0.5[0.93 \cdot 0.98ln(c_{2Hh}) + 0.93 \cdot 0.02ln(c_{2Hl}) + 0.07 \cdot 0.98ln(c_{2Lh}) + 0.07 \cdot 0.02ln(c_{2Ll})]$$
(A.1)
$$-\lambda(x+y+u-1) - \mu_x x - \mu_y y - \mu_u u,$$

with
$$c_1 = x + yp^{**}$$
,
 $c_{2Hh} = 1.54u + \left(\frac{x}{p^{**}} + y\right) 1.3$,
 $c_{2Hl} = 1.54u + \left(\frac{x}{p^{**}} + y\right) 0.3$,
 $c_{2Lh} = 0.25u + \left(\frac{x}{p^{**}} + y\right) 1.3$,
 $c_{2Ll} = 0.25u + \left(\frac{x}{p^{**}} + y\right) 0.3$,

where λ is the Lagrange multiplier corresponding to the budget constraint (11) and μ_x, μ_y and μ_u are the Lagrange multipliers corresponding to the non-negativity conditions (12). Considering that $p^{**} = 1$ (see Section 4) banks equally split their investment in liquid assets in order to hedge their idiosyncratic liquidity risks, so we have $x^* = y^* = 0.5z^*$ (for a detailed explanation see Section 5.1). Differentiating \mathcal{L} with respect to z, u, λ, μ_z and μ_u gives

$$\begin{aligned} \frac{\partial \mathcal{L}}{\partial z} &= \frac{0.5}{z} + \frac{0.5 \cdot 0.93 \cdot 0.98 \cdot 1.3}{1.3z + 1.54u} + \frac{0.5 \cdot 0.93 \cdot 0.02 \cdot 0.3}{0.3z + 1.54u} \\ &+ \frac{0.5 \cdot 0.07 \cdot 0.98 \cdot 1.3}{1.3z + 0.25u} + \frac{0.5 \cdot 0.07 \cdot 0.02 \cdot 0.3}{0.3z + 0.25u} - \lambda - \mu_z \stackrel{!}{=} 0, \end{aligned} \tag{A.2}$$

$$\begin{aligned} \frac{\partial \mathcal{L}}{\partial u} &= \frac{0.5 \cdot 0.93 \cdot 0.98 \cdot 1.54}{1.3z + 1.54u} + \frac{0.5 \cdot 0.93 \cdot 0.02 \cdot 1.54}{0.3z + 1.54u} \\ &+ \frac{0.5 \cdot 0.07 \cdot 0.98 \cdot 0.25}{1.3z + 0.25u} + \frac{0.5 \cdot 0.07 \cdot 0.02 \cdot 0.25}{0.3z + 0.25u} - \lambda - \mu_u \stackrel{!}{=} 0, \end{aligned} \tag{A.3}$$

$$\frac{\partial \mathcal{L}}{\partial \lambda} = 1 - z - u \stackrel{!}{=} 0, \tag{A.4}$$

$$\frac{\partial \mathcal{L}}{\partial \mu_z} = -z \stackrel{!}{=} 0,\tag{A.5}$$

$$\frac{\partial \mathcal{L}}{\partial \mu_u} = -u \stackrel{!}{=} 0. \tag{A.6}$$

Multiplying both sides of equation (A.2) with z, of equation (A.3) with u, adding the two equations and regarding equation (A.4), we obtain $\lambda = 1$. Testing whether a non-negativity constraint binds, reveals that this constraint binds for u, so that $u^* = 0$ and hence $\mu_u \neq 0$. Considering the constraint (11) and $u^* = 0$, induces that $z^* = 1$, i.e. the representative bank invests its total amount of deposits in liquid assets.

Proof II. When equity capital is available for banks their optimisation problem reads

$$\max_{x,y,u,e_{2Hh},e_{2Lh},e_{2Ll}} \mathcal{L} = 0.5ln(c_1) + 0.5[0.93 \cdot 0.98ln(c_{2Hh}) + 0.93 \cdot 0.02ln(c_{2Hl}) + 0.07 \cdot 0.98ln(c_{2Lh}) + 0.07 \cdot 0.02ln(c_{2Ll})] - \lambda \left(x + y + u \right) - 1 - \left[\frac{0.5}{1.5} (1.4497u + 0.9114e_{2Hh} + 0.0186e_{2Hl}) + 0.0686e_{2Lh} + 0.0014e_{2Ll}) \right] - \mu_{x}x - \mu_{y}y - \mu_{u}u - \mu_{e_{2Hh}}e_{2Hh} - \mu_{e_{2Hl}}e_{2Hl} - \mu_{e_{2Lh}}e_{2Lh} - \mu_{e_{2Ll}}e_{2Ll},$$
(A.7)

with
$$c_1 = x + yp^{**}$$
,
 $c_{2Hh} = 1.54u + \left(\frac{x}{p^{**}} + y\right) 1.3 - e_{2Hh}$,
 $c_{2Hl} = 1.54u + \left(\frac{x}{p^{**}} + y\right) 0.3 - e_{2Hl}$,
 $c_{2Lh} = 0.25u + \left(\frac{x}{p^{**}} + y\right) 1.3 - e_{2Lh}$,
 $c_{2Ll} = 0.25u + \left(\frac{x}{p^{**}} + y\right) 0.3 - e_{2Ll}$,

where λ is the Lagrange multiplier corresponding to the budget constraint (11). We capture the investors' incentive-compatibility constraint (9) by respectively replacing e_0 in the budget constraint. The variables $\mu_x, \mu_y, \mu_u, \mu_{e_{2Hh}}, \mu_{e_{2Lh}}, \mu_{e_{2Lh}}, \mu_{e_{2Ll}}$ are the Lagrange multipliers corresponding to the non-negativity conditions (12). As the same argumentation holds as in sections 4 and 5.1, we have $p^{**} = 1$ and $x^* = y^* = 0.5z^*$. By differentiating the Lagrange function with respect to $z, u, e_{2Hh}, e_{2Hl}, e_{2Lh}, \mu_z, \mu_u, \lambda, \mu_{e_{2Hh}}, \mu_{e_{2Hl}}, \mu_{e_{2Lh}}$ and $\mu_{e_{2Lh}}$, we obtain

$$\begin{aligned} \frac{\partial \mathcal{L}}{\partial z} &= \frac{0.5}{z} + \frac{0.5 \cdot 0.93 \cdot 0.98 \cdot 1.3}{1.3z + 1.54u - e_{2Hh}} + \frac{0.5 \cdot 0.93 \cdot 0.02 \cdot 0.3}{0.3z + 1.54u - e_{2Hl}} \\ &+ \frac{0.5 \cdot 0.07 \cdot 0.98 \cdot 1.3}{1.3z + 0.25u - e_{2Lh}} + \frac{0.5 \cdot 0.07 \cdot 0.02 \cdot 0.3}{0.3z + 0.25u - e_{2Ll}} - \lambda - \mu_z \stackrel{!}{=} 0, \end{aligned}$$
(A.8)
$$\frac{\partial \mathcal{L}}{\partial u} &= \frac{0.5 \cdot 0.93 \cdot 0.98 \cdot 1.54}{1.3z + 1.54u - e_{2Hh}} + \frac{0.5 \cdot 0.93 \cdot 0.02 \cdot 1.54}{0.3z + 1.54u - e_{2Hl}} \\ &+ \frac{0.5 \cdot 0.07 \cdot 0.98 \cdot 0.25}{1.2z + 0.25u - e_{2Lh}} + \frac{0.5 \cdot 0.07 \cdot 0.02 \cdot 0.25}{0.2z + 1.54u - e_{2Hl}} \end{aligned}$$
(A.8)

$$\frac{\partial \mathcal{L}}{\partial \mathcal{L}} = \frac{0.5 \cdot 0.93 \cdot 0.98 \cdot (-1)}{1.5} - \lambda \left(-\frac{0.5}{1.5} \cdot 0.9114 \right) - \mu_{e_{0.00}} \stackrel{!}{=} 0, \tag{A.10}$$

$$\frac{\partial e_{2Hh}}{\partial e_{2Hh}} = \frac{1.3z + 1.54u - e_{2Hh}}{1.3z + 1.54u - e_{2Hh}} - \lambda \left(-\frac{1.5}{1.5} \cdot 0.9114\right) - \mu_{e_{2Hh}} \equiv 0, \tag{A.10}$$

$$\frac{\partial \mathcal{L}}{\partial e_{2Hh}} = \frac{0.5 \cdot 0.93 \cdot 0.02 \cdot (-1)}{0.22 \cdot 0.012} - \lambda \left(-\frac{0.5}{1.5} \cdot 0.0186\right) - \mu_{e_{2Hh}} \stackrel{!}{=} 0, \tag{A.11}$$

$$\frac{\partial e_{2Hl}}{\partial e_{2Lh}} = \frac{0.5 \cdot 0.07 \cdot 0.98 \cdot (-1)}{1.3z + 0.25u - e_{2Lh}} - \lambda \left(-\frac{0.5}{1.5} \cdot 0.0686 \right) - \mu_{e_{2Lh}} \stackrel{!}{=} 0, \tag{A.12}$$

$$\frac{\partial \mathcal{L}}{\partial e_{2Ll}} = \frac{0.5 \cdot 0.07 \cdot 0.02 \cdot (-1)}{0.3z + 0.25u - e_{2Ll}} - \lambda \left(-\frac{0.5}{1.5} \cdot 0.0014 \right) - \mu_{e_{2Ll}} \stackrel{!}{=} 0, \tag{A.13}$$

$$\frac{\partial \mathcal{L}}{\partial \lambda} = z + u - 1 - \left[\frac{0.5}{1.5}(1.4497u + 0.9114e_{2Hh} + 0.0186e_{2Hl} + 0.0686e_{2Lh} + 0.0014e_{2Ll})\right] \stackrel{!}{=} 0,$$
(A.14)

$$\frac{\partial \mathcal{L}}{\partial \mu_z} = -z \stackrel{!}{=} 0, \tag{A.15}$$

$$\frac{\partial \mathcal{L}}{\partial \mu_u} = -u \stackrel{!}{=} 0, \tag{A.16}$$

$$\frac{\partial \mathcal{L}}{\partial \mu_{e_{2Hh}}} = -e_{2Hh} \stackrel{!}{=} 0, \tag{A.17}$$

$$\frac{\partial \mathcal{L}}{\partial \mu_{e_{2Hl}}} = -e_{2Hl} \stackrel{!}{=} 0, \tag{A.18}$$

$$\frac{\partial \mathcal{L}}{\partial \mu_{e_{2Lh}}} = -e_{2Lh} \stackrel{!}{=} 0, \tag{A.19}$$

$$\frac{\partial \mathcal{L}}{\partial \mu_{e_{2Ll}}} = -e_{2Ll} \stackrel{!}{=} 0. \tag{A.20}$$

Multiplying both sides of equation (A.8) with z, of (A.9) with u, of (A.10) with e_{2Hh} , of (A.11) with e_{2Hl} , of (A.12) with e_{2Lh} and of (A.13) with e_{2Ll} , adding the six equations and regarding equation (A.14), we again obtain $\lambda = 1$. After testing which non-negativity conditions bind, we derive that the non-negativity conditions for e_{Hh} , e_{Hl} , e_{Ll} and e_{Ll} become binding, i.e. $e_{Hh}^* = e_{Hl}^* = e_{Ll}^* = e_{Ll}^* = 0$ and thus $\mu_{e_{2Hh}} = \mu_{e_{2Hl}} = \mu_{e_{2Lh}} = \mu_{e_{2Ll}} \neq$ 0. Solving then for z^* and u^* we get $z^* = 0.9088$ and $u^* = 0.1765$ and regarding the constraint (9) the optimal amount of equity capital is $e_0^* = 0.0853$.

Proof III. When a bank faces capital requirements for loans $(CR^{min} = 0.5799 = \frac{e_0}{u})$, its optimisation problem can be formulated as

$$\max_{x,y,u,e_{2Hh},e_{2Lh},e_{2Ll}} \mathcal{L} = 0.5ln(c_1) + 0.5[0.93 \cdot 0.98ln(c_{2Hh}) + 0.93 \cdot 0.02ln(c_{2Hl}) + 0.07 \cdot 0.98ln(c_{2Lh}) + 0.07 \cdot 0.02ln(c_{2Ll})] -\lambda \Big(x + y + u - 1 - \left[\frac{0.5}{1.5} (1.4497u + 0.9114e_{2Hh} + 0.0186e_{2Hl} + 0.0686e_{2Lh} + 0.0014e_{2Ll}) \right] \Big)$$
(A.21)
$$-\mu_x x - \mu_y y - \mu_u u - \mu_{e_{2Hh}}e_{2Hh} -\mu_{e_{2Hl}}e_{2Hl} - \mu_{e_{2Lh}}e_{2Lh} - \mu_{e_{2Lh}}e_{2Ll} - \mu_{CR} \left(\frac{0.5}{1.5} \frac{1}{u} \Big[1.4497u + 0.9114e_{2Hh} + 0.0186e_{2Hl} + 0.0686e_{2Lh} + 0.0014e_{2Ll} \Big] - 0.5799 \Big),$$

with $c_1 = x + yp^{**}$, $c_{2Hh} = 1.54u + \left(\frac{x}{p^{**}} + y\right) 1.3 - e_{2Hh}$, $c_{2Hl} = 1.54u + \left(\frac{x}{p^{**}} + y\right) 0.3 - e_{2Hl}$, $c_{2Lh} = 0.25u + \left(\frac{x}{p^{**}} + y\right) 1.3 - e_{2Lh}$, $c_{2Ll} = 0.25u + \left(\frac{x}{p^{**}} + y\right) 0.3 - e_{2Ll}$,

where λ is the Lagrange multiplier corresponding to the budget constraint (11). We capture the investors' incentive-compatibility constraint (9) by respectively replacing e_0 in the budget constraint and in the regulation constraint. The variables $\mu_x, \mu_y, \mu_u, \mu_{e_{2Hh}}, \mu_{e_{2Hl}}, \mu_{e_{2Lh}}$ and $\mu_{e_{2Ll}}$ are the Lagrange multipliers corresponding to the non-negativity conditions (12) and μ_{CR} is the Lagrange multiplier corresponding to the regulation constraint (10). Considering that $p^{**} = 1$ (see Section 4) as well as $x^* = y^* = 0.5z^*$ (for a detailed explanation see Section 5.1) and differentiating \mathcal{L} with respect to $z, u, e_{2Hh}, e_{2Hl}, e_{2Lh}, e_{2Ll}, \lambda, \mu_{CR}, \mu_{e_{2Hh}}, \mu_{e_{2Ll}}$ and $\mu_{e_{2Ll}}$ gives

$$\begin{aligned} \frac{\partial \mathcal{L}}{\partial z} &= \frac{0.5}{z} + \frac{0.5 \cdot 0.93 \cdot 0.98 \cdot 1.3}{1.3z + 1.54u - e_{2Hh}} + \frac{0.5 \cdot 0.93 \cdot 0.02 \cdot 0.3}{0.3z + 1.54u - e_{2Hl}} \\ &+ \frac{0.5 \cdot 0.07 \cdot 0.98 \cdot 1.3}{1.3z + 0.25u - e_{2Lh}} + \frac{0.5 \cdot 0.07 \cdot 0.02 \cdot 0.3}{0.3z + 0.25u - e_{2Ll}} - \lambda - \mu_z \stackrel{!}{=} 0, \end{aligned}$$

$$\begin{aligned} \frac{\partial \mathcal{L}}{\partial u} &= \frac{0.5 \cdot 0.93 \cdot 0.98 \cdot 1.54}{1.3z + 1.54u - e_{2Hh}} + \frac{0.5 \cdot 0.93 \cdot 0.02 \cdot 1.54}{0.3z + 1.54u - e_{2Hl}} \\ &+ \frac{0.5 \cdot 0.07 \cdot 0.98 \cdot 0.25}{1.3z + 0.25u - e_{2Lh}} + \frac{0.5 \cdot 0.07 \cdot 0.02 \cdot 0.25}{0.3z + 1.54u - e_{2Hl}} \\ &+ \frac{0.5 \cdot 0.07 \cdot 0.98 \cdot 0.25}{1.3z + 0.25u - e_{2Lh}} + \frac{0.5 \cdot 0.07 \cdot 0.02 \cdot 0.25}{0.3z + 0.25u - e_{2Ll}} \\ &- \lambda \left(1 - \left(\frac{0.5 \cdot 1.4497}{1.5} \right) \right) + \mu_{CR} \left(\frac{\frac{0.5}{1.5}}{u^2} [0.9114e_{2Hh} \right) \\ &+ 0.0186e_{2Hl} + 0.0686e_{2Lh} + 0.0014e_{2Ll} \right) - \mu_u \stackrel{!}{=} 0, \end{aligned}$$

$$\begin{aligned} \frac{\partial \mathcal{L}}{\partial e_{2Hh}} &= \frac{0.5 \cdot 0.93 \cdot 0.98 \cdot (-1)}{1.3z + 1.54u - e_{2Hl}} - \lambda \left(-\frac{0.5}{1.5} \cdot 0.9114 \right) - \mu_{e_{2Hh}} \\ &- \mu_{CR} \left(\frac{\frac{0.5}{1.5} \cdot 0.9114}{u} \right) \stackrel{!}{=} 0, \end{aligned}$$

$$(A.24) \\ \frac{\partial \mathcal{L}}{\partial e_{2Hl}} &= \frac{0.5 \cdot 0.93 \cdot 0.02 \cdot (-1)}{0.3z + 1.54u - e_{2Hl}} - \lambda \left(-\frac{0.5}{1.5} \cdot 0.0186 \right) - \mu_{e_{2Hl}} \\ &- \mu_{CR} \left(\frac{\frac{0.5}{1.5} \cdot 0.0186}{u} \right) \stackrel{!}{=} 0, \end{aligned}$$

$$\frac{\partial \mathcal{L}}{\partial e_{2Lh}} = \frac{0.5 \cdot 0.07 \cdot 0.98 \cdot (-1)}{1.3z + 0.25u - e_{2Lh}} - \lambda \left(-\frac{0.5}{1.5} \cdot 0.0686 \right) - \mu_{e_{2Lh}} - \mu_{CR} \left(\frac{\frac{0.5}{1.5} \cdot 0.0686}{u} \right) \stackrel{!}{=} 0,$$
(A.26)

$$\frac{\partial \mathcal{L}}{\partial e_{2Ll}} = \frac{0.5 \cdot 0.07 \cdot 0.02 \cdot (-1)}{0.3z + 0.25u - e_{2Ll}} - \lambda \left(-\frac{0.5}{1.5} \cdot 0.0014 \right) - \mu_{e_{2Ll}} - \mu_{CR} \left(\frac{0.5}{1.5} \cdot 0.0014 \right) \stackrel{!}{=} 0,$$
(A.27)

$$\frac{\partial \mathcal{L}}{\partial \lambda} = z + u - 1 - \left[\frac{0.5}{1.5}(1.4497u + 0.9114e_{2Hh} + 0.0186e_{2Hl} + 0.0686e_{2Lh} + 0.0014e_{2Ll})\right] \stackrel{!}{=} 0.$$
(A.28)

$$\frac{\partial \mathcal{L}}{\partial \mu_{CR}} = \frac{\frac{0.5}{1.5}}{u} \left[1.4497u + 0.9114e_{2Hh} + 0.0186e_{2Hl} + 0.0686e_{2Lh} + 0.0014e_{2Ll} \right] - 0.5799 \stackrel{!}{=} 0.$$
(A.29)

$$\frac{\partial \mathcal{L}}{\partial \mu_z} = -z \stackrel{!}{=} 0, \tag{A.30}$$

$$\frac{\partial \mathcal{L}}{\partial \mu_u} = -u \stackrel{!}{=} 0,\tag{A.31}$$

$$\frac{\partial \mathcal{L}}{\partial \mu_{e_{2Hh}}} = -e_{2Hh} \stackrel{!}{=} 0, \tag{A.32}$$

$$\frac{\partial \mathcal{L}}{\partial \mu_{e_{2Hl}}} = -e_{2Hl} \stackrel{!}{=} 0, \tag{A.33}$$

$$\frac{\partial \mathcal{L}}{\partial \mu_{e_{2Lh}}} = -e_{2Lh} \stackrel{!}{=} 0, \tag{A.34}$$

$$\frac{\partial \mathcal{L}}{\partial \mu_{e_{2Ll}}} = -e_{2Ll} \stackrel{!}{=} 0. \tag{A.35}$$

Multiplying both sides of equation (A.22) with z, of (A.23) with u, of (A.24) with e_{2Hh} , of (A.25) with e_{2Hl} , of (A.26) with e_{2Lh} and of (A.27) with e_{2Ll} , adding the six equations and regarding equation (A.28), we again obtain $\lambda = 1$. After testing which non-negativity conditions bind, we derive that the non-negativity conditions for e_{Hl}, e_{Ll} and e_{Ll} become binding, i.e. $= e_{Hl}^* = e_{Ll}^* = e_{Ll}^* = 0$ and thus $\mu_{e_{2Hl}} = \mu_{e_{2Lh}} = \mu_{e_{2Ll}} \neq 0$. Solving then for z^* , u^* and e_{2Hh}^* we get $z^* = 0.9162$, $u^* = 0.1995$ and $e_{2Hh}^* = 0.0635$. Regarding constraint (9), the optimal amount of equity capital is $e_0^* = 0.1157$.

Proof IV. When banks face capital requirements for loans and government bonds $(CR^{min} = 0.5799 = \frac{e_0}{u+0.05y})$, their optimisation problem becomes

$$\max_{x,y,u,e_{2Hh},e_{2Lh},e_{2Ll}} \mathcal{L} = 0.5ln(c_1) + 0.5[0.93 \cdot 0.98ln(c_{2Hh}) + 0.93 \cdot 0.02ln(c_{2Hl}) + 0.07 \cdot 0.98ln(c_{2Lh}) + 0.07 \cdot 0.02ln(c_{2Ll})] -\lambda \Big(x + y + u - 1 - \Big[\frac{0.5}{1.5} (1.4497u + 0.9114e_{2Hh} + 0.0186e_{2Hl} + 0.0686e_{2Lh} + 0.0014e_{2Ll}) \Big] \Big) -\mu_x x - \mu_y y - \mu_u u - \mu_{e_{2Hh}}e_{2Hh} -\mu_{e_{2Hl}}e_{2Hl} - \mu_{e_{2Lh}}e_{2Lh} - \mu_{e_{2Lh}}e_{2Ll} - \mu_{CR} \Big(\frac{0.5}{1.5} \frac{1}{u + 0.05y} \Big[1.4497u + 0.9114e_{2Hh} + 0.0186e_{2Hl} + 0.0686e_{2Lh} + 0.0014e_{2Ll} \Big] - 0.5799 \Big),$$

with
$$c_1 = x + yp^{**}$$
,
 $c_{2Hh} = 1.54u + \left(\frac{x}{p^{**}} + y\right) 1.3 - e_{2Hh}$,
 $c_{2Hl} = 1.54u + \left(\frac{x}{p^{**}} + y\right) 0.3 - e_{2Hl}$,
 $c_{2Lh} = 0.25u + \left(\frac{x}{p^{**}} + y\right) 1.3 - e_{2Lh}$,
 $c_{2Ll} = 0.25u + \left(\frac{x}{p^{**}} + y\right) 0.3 - e_{2Ll}$,

where λ is the Lagrange multiplier corresponding to the budget constraint (11). We capture the investors' incentive-compatibility constraint (9) by respectively replacing e_0 in the budget constraint and in the regulation constraint. The variables $\mu_x, \mu_y, \mu_u, \mu_{e_{2Hh}}, \mu_{e_{2Lh}}, \mu_{e_{2Ll}}$ are the Lagrange multipliers corresponding to the nonnegativity conditions (12) and μ_{CR} is the Lagrange multiplier corresponding to the regulation constraint (10). Considering that $p^{**} = 1$ (see Section 4) banks equally split their investment in liquid assets ($x^* = y^* = 0.5z^*$) also when sovereign bonds are subject to capital regulation. By differentiating \mathcal{L} with resect to $z, u, e_{2Hh}, e_{2Lh}, e_{2Lh}, \lambda, \mu_{CR},$ $\mu_{e_{2Hh}}, \mu_{e_{2Lh}}, \mu_{e_{2Lh}}$ and $\mu_{e_{2Ll}}$ we obtain

$$\frac{\partial \mathcal{L}}{\partial z} = \frac{0.5}{z} + \frac{0.5 \cdot 0.93 \cdot 0.93 \cdot 1.3}{1.3z + 1.54u - e_{2Hh}} + \frac{0.5 \cdot 0.93 \cdot 0.02 \cdot 0.3}{0.3z + 1.54u - e_{2Hl}} \\
+ \frac{0.5 \cdot 0.07 \cdot 0.98 \cdot 1.3}{1.3z + 0.25u - e_{2Lh}} + \frac{0.5 \cdot 0.07 \cdot 0.02 \cdot 0.3}{0.3z + 0.25u - e_{2Ll}} - \lambda - \mu_z \\
+ \mu_{CR} \left(\frac{\frac{0.5 \cdot 0.025}{1.5}}{(u + 0.025z)^2} [1.4497u + 0.9114e_{2Hh} + 0.0186e_{2Hl} + 0.0686e_{2Lh} + 0.0014e_{2Ll}] \right) \stackrel{!}{=} 0, \\
\frac{\partial \mathcal{L}}{\partial u} = \frac{0.5 \cdot 0.93 \cdot 0.98 \cdot 1.54}{1.3z + 1.54u - e_{2Hh}} + \frac{0.5 \cdot 0.93 \cdot 0.02 \cdot 1.54}{0.3z + 1.54u - e_{2Hl}} \\
+ \frac{0.5 \cdot 0.07 \cdot 0.98 \cdot 0.25}{1.3z + 0.25u - e_{2Lh}} + \frac{0.5 \cdot 0.07 \cdot 0.02 \cdot 0.25}{0.3z + 0.25u - e_{2Ll}} \\
- \lambda \left(1 - \left(\frac{0.5 \cdot 1.4497}{1.5} \right) \right) + \mu_{CR} \left(\frac{\frac{0.5}{1.5}}{(u + 0.025z)^2} \right)$$
(A.38)

 $[1.4497 \cdot 0.025z + 0.9114e_{2Hh} + 0.0186e_{2Hl}$

$$+ 0.0686e_{2Lh} + 0.0014e_{2Ll}] - \mu_u \stackrel{!}{=} 0,$$

$$\frac{\partial \mathcal{L}}{\partial e_{2Hh}} = \frac{0.5 \cdot 0.93 \cdot 0.98 \cdot (-1)}{1.3z + 1.54u - e_{2Hh}} - \lambda \left(-\frac{0.5}{1.5} \cdot 0.9114 \right) - \mu_{e_{2Hh}}$$

$$- \mu_{CR} \left(\frac{0.3038}{u + 0.025z} \right) \stackrel{!}{=} 0,$$

$$(A.39)$$

$$\frac{\partial \mathcal{L}}{\partial e_{2Hl}} = \frac{0.5 \cdot 0.93 \cdot 0.02 \cdot (-1)}{0.3z + 1.54u - e_{2Hl}} - \lambda \left(-\frac{0.5}{1.5} \cdot 0.0186 \right) - \mu_{e_{2Hl}} - \mu_{CR} \left(\frac{0.062}{1 + 0.025} \right) \stackrel{!}{=} 0, \tag{A.40}$$

$$\frac{\partial \mathcal{L}}{\partial e_{2Lh}} = \frac{0.5 \cdot 0.07 \cdot 0.98 \cdot (-1)}{1.3z + 0.25u - e_{2Lh}} - \lambda \left(-\frac{0.5}{1.5} \cdot 0.0686 \right) - \mu_{e_{2Lh}} - \mu_{CR} \left(\frac{0.02286}{u + 0.025z} \right) \stackrel{!}{=} 0,$$
(A.41)

$$\frac{\partial \mathcal{L}}{\partial e_{2Ll}} = \frac{0.5 \cdot 0.07 \cdot 0.02 \cdot (-1)}{0.3z + 0.25u - e_{2Ll}} - \lambda \left(-\frac{0.5}{1.5} \cdot 0.0014 \right) - \mu_{e_{2Ll}} - \mu_{CR} \left(\frac{0.00046}{u + 0.025z} \right) \stackrel{!}{=} 0,$$
(A.42)

$$\frac{\partial \mathcal{L}}{\partial \lambda} = z + u - 1 - \left[\frac{0.5}{1.5}(1.4497u + 0.9114e_{2Hh} + 0.0186e_{2Hl} + 0.0686e_{2Lh} + 0.0014e_{2Ll})\right] \stackrel{!}{=} 0.$$
(A.43)

$$\frac{\partial \mathcal{L}}{\partial \mu_{CR}} = \frac{\frac{0.5}{1.5}}{u + 0.025z} \left[1.4497u + 0.9114e_{2Hh} + 0.0186e_{2Hl} + 0.0686e_{2Lh} + 0.0014e_{2Ll} \right] - 0.5799 \stackrel{!}{=} 0.$$
(A.44)

$$\frac{\partial \mathcal{L}}{\partial \mu_z} = -z \stackrel{!}{=} 0,\tag{A.45}$$

$$\frac{\partial \mathcal{L}}{\partial \mu_u} = -u \stackrel{!}{=} 0, \tag{A.46}$$

$$\frac{\partial \mathcal{L}}{\partial \mu_{e_{2Hh}}} = -e_{2Hh} \stackrel{!}{=} 0, \tag{A.47}$$

$$\frac{\partial \mathcal{L}}{\partial \mu_{e_{2Hl}}} = -e_{2Hl} \stackrel{!}{=} 0, \tag{A.48}$$

$$\frac{\partial \mathcal{L}}{\partial \mu_{e_{2Lh}}} = -e_{2Lh} \stackrel{!}{=} 0, \tag{A.49}$$

$$\frac{\partial \mathcal{L}}{\partial \mu_{e_{2Ll}}} = -e_{2Ll} \stackrel{!}{=} 0, \tag{A.50}$$

Multiplying both sides of equation (A.37) with z, of (A.38) with u, of (A.39) with e_{2Hh} , of (A.40) with e_{2Hl} , of (A.41) with e_{2Lh} and of (A.42) with e_{2Ll} , adding the 6 equations and regarding equation (A.43), we again obtain $\lambda = 1$. After testing which non-negativity conditions bind, we derive that the non-negativity conditions for e_{Hl} , e_{Ll} and e_{Ll} become binding, i.e. $= e_{Hl}^* = e_{Ll}^* = e_{Ll}^* = 0$ and thus $\mu_{e_{2Hl}} = \mu_{e_{2Lh}} = \mu_{e_{2Ll}} \neq 0$. Solving then for z^* , u^* and e_{2Hh}^* , we get $z^* = 0.9172$, $u^* = 0.2287$ and $e_{2Hh}^* = 0.1166$. Inserting z^* , u^* and e_{2Hh}^* in constraint (9), the optimal amount of equity capital a bank raises is $e_0^* = 0.1459$.

Bibliography

- Acharya, V. V., I. Drechsler, and P. Schnabl (2014). A pyrrhic victory? Bank bailouts and sovereign credit risk. *The Journal of Finance* 69(6), 2689–2739.
- Acharya, V. V. and R. G. Rajan (2013). Sovereign debt, government myopia, and the financial sector. *The Review of Financial Studies* 26(6), 1526–1560.
- Acharya, V. V. and S. Steffen (2015). The "greatest" carry trade ever? Understanding eurozone bank risks. Journal of Financial Economics 115(2), 215–236.
- Allen, F. and E. Carletti (2006). Credit risk transfer and contagion. Journal of Monetary Economics 53(1), 89–111.
- Allen, F. and D. Gale (2000). Financial contagion. Journal of Political Economy 108(1), 1–33.

- Allen, F. and D. Gale (2006). Systemic risk and regulation. In M. Carey and R. M. Stulz (Eds.), The Risks of Financial Institutions, pp. 341–376. Chicago: Chicago University Press.
- Altavilla, C., M. Pagano, and S. Simonelli (2017). Bank exposures and sovereign stress transmission. *Review of Finance* 21(6), 2103–2139.
- Alter, A. and A. Beyer (2014). The dynamics of spillover effects during the European sovereign debt turmoil. Journal of Banking & Finance 42, 134–153.
- Angelini, P., G. Grande, and F. Panetta (2014). The negative feedback loop between banks and sovereigns. Bank of Italy Occasional Paper No. 213.
- Bagehot, W. (1873). Lombard Street: A description of the money market. New York: Scribner, Armstrong & Company.
- Blum, J. M. (1999). Do capital adequacy requirements reduce risks in banking? Journal of Banking & Finance 23(5), 755–771.
- Bolton, P. and O. Jeanne (2011). Sovereign default risk and bank fragility in financially integrated economies. *IMF Economic Review* 59(2), 162–194.
- Broner, F., A. Erce, A. Martin, and J. Ventura (2014). Sovereign debt markets in turbulent times: Creditor discrimination and crowding-out effects. *Journal of Monetary Economics 61*, 114–142.
- Brunnermeier, M. K., L. Garicano, P. R. Lane, M. Pagano, R. Reis, T. Santos, D. Thesmar, S. van Nieuwerburgh, and D. Vayanos (2016). The sovereign-bank diabolic loop and ESBies. *The American Economic Review* 106(5), 508–512.
- Cooper, R. and K. Nikolov (2013). Government debt and banking fragility: The spreading of strategic uncertainty. NBER Working Paper Series, No. 19278.
- Corsetti, G., K. Kuester, A. Meier, and G. J. Müller (2013). Sovereign risk, fiscal policy, and macroeconomic stability. *The Economic Journal* 123(566), 99–132.

- Covi, G. and U. Eydam (2016). End of the sovereign-bank doom loop in the European Union? The bank recovery and resolution directive. Kiel Advanced Studies Working Paper Series, No. 468.
- De Bruyckere, V., M. Gerhardt, G. Schepens, and R. Vander Vennet (2013). Bank/sovereign risk spillovers in the European debt crisis. *Journal of Banking & Finance 37*(12), 4793–4809.
- Diamond, D. W. and P. H. Dybvig (1983). Bank runs, deposit insurance, and liquidity. Journal of Political Economy 91(3), 401–419.
- Ejsing, J. W. and W. Lemke (2011). The Janus-headed salvation: Sovereign and bank credit risk premia during 2008-2009. *Economic Letters* 110(1), 28–31.
- European Systemic Risk Board (2015). ESRB report on the regulatory treatment of sovereign exposures. Available at http://www.esrb.europa.eu/pub/pdf/ other/esrbreportregulatorytreatmentsovereignexposures032015.en.pdf? c0cad80cf39a74e20d9d5947c7390df1. Accessed: 2015-05-18.
- Farhi, E. and J. Tirole (2016). Deadly embrace: Sovereign and financial balance sheets doom loops. NBER Working Paper Series, No. 21843.
- Flannery, M. J. (1989). Capital regulation and insured banks choice of individual loan default risks. *Journal of Monetary Economics* 24 (2), 235–258.
- Furlong, F. T. and M. C. Keeley (1989). Capital regulation and bank risk-taking: A note. Journal of Banking & Finance 13(6), 883–891.
- Gennaioli, N., A. Martin, and S. Rossi (2014a). Banks, government bonds, and default: What do the data say? CEPR Discussion Paper Series, No. 10044.
- Gennaioli, N., A. Martin, and S. Rossi (2014b). Sovereign default, domestic banks, and financial institutions. *The Journal of Finance* 69(2), 819–866.
- Gross, M. and C. Kok (2013). Measuring contagion potential among sovereigns and banks using a mixed-cross-section GVAR. ECB Working Paper Series, No. 1570.

- Harris, M., C. C. Opp, and M. M. Opp (2014). Higher capital requirements, safer banks? Macroprudential regulation in a competitive financial system. Mimeo.
- Haubrich, J. G. and P. Wachtel (1993). Capital requirements and shifts in commercial bank portfolios. *Federal Reserve Bank of Cleveland Economic Review 29*(3), 2–15.
- Heyde, F. and U. Neyer (2010). Credit default swaps and the stability of the banking sector. International Review of Finance 10(1), 27–61.
- Hyun, J.-S. and B.-K. Rhee (2011). Bank capital regulation and credit supply. Journal of Banking & Finance 35(2), 323–330.
- Korte, J. and S. Steffen (2015). Zero risk contagion-banks' sovereign exposure and sovereign risk spillovers. Mimeo.
- Vergote, O. (2016). Credit risk spillover between financials and sovereigns in the euro area during 2007-2015. ECB Working Paper Series, No. 1898.
- Weidmann, J. (2016). Dinner speach at the 6th Frankfurt Finance Summit, Frankfurt Main, May 5th. Available at https://www.bundesbank.de/Redaktion/EN/Reden/ 2016/2016_05_11_weidmann.html. Accessed: 2017-06-11.

PREVIOUS DISCUSSION PAPERS

- 275 Neyer, Ulrike and Sterzel, André, Capital Requirements for Government Bonds Implications for Bank Behaviour and Financial Stability, December 2017.
- 274 Deckers, Thomas, Falk, Armin, Kosse, Fabian, Pinger, Pia and Schildberg-Hörisch, Hannah, Socio-Economic Status and Inequalities in Children's IQ and Economic Preferences, November 2017.
- 273 Defever, Fabrice, Fischer, Christian and Suedekum, Jens, Supplier Search and Re-matching in Global Sourcing – Theory and Evidence from China, November 2017.
- 272 Thomas, Tobias, Heß, Moritz and Wagner, Gert G., Reluctant to Reform? A Note on Risk-Loving Politicians and Bureaucrats, October 2017.
- 271 Caprice, Stéphane and Shekhar, Shiva, Negative Consumer Value and Loss Leading, October 2017.
- 270 Emch, Eric, Jeitschko, Thomas D. and Zhou, Arthur, What Past U.S. Agency Actions Say About Complexity in Merger Remedies, With an Application to Generic Drug Divestitures, October 2017.
- 269 Goeddeke, Anna, Haucap, Justus, Herr, Annika and Wey, Christian, Flexibility in Wage Setting Under the Threat of Relocation, September 2017. Forthcoming in: Labour: Review of Labour Economics and Industrial Relations.
- 268 Haucap, Justus, Merger Effects on Innovation: A Rationale for Stricter Merger Control?, September 2017. Forthcoming in: Concurrences: Competition Law Review.
- 267 Brunner, Daniel, Heiss, Florian, Romahn, André and Weiser, Constantin, Reliable Estimation of Random Coefficient Logit Demand Models, September 2017.
- 266 Kosse, Fabian, Deckers, Thomas, Schildberg-Hörisch, Hannah and Falk, Armin, The Formation of Prosociality: Causal Evidence on the Role of Social Environment, July 2017.
- 265 Friehe, Tim and Schildberg-Hörisch, Hannah, Predicting Norm Enforcement: The Individual and Joint Predictive Power of Economic Preferences, Personality, and Self-Control, July 2017.
- 264 Friehe, Tim and Schildberg-Hörisch, Hannah, Self-Control and Crime Revisited: Disentangling the Effect of Self-Control on Risk Taking and Antisocial Behavior, July 2017.
- 263 Golsteyn, Bart and Schildberg-Hörisch, Hannah, Challenges in Research on Preferences and Personality Traits: Measurement, Stability, and Inference, July 2017.
- 262 Lange, Mirjam R.J., Tariff Diversity and Competition Policy Drivers for Broadband Adoption in the European Union, July 2017.
- 261 Reisinger, Markus and Thomes, Tim Paul, Manufacturer Collusion: Strategic Implications of the Channel Structure, July 2017.
- 260 Shekhar, Shiva and Wey, Christian, Uncertain Merger Synergies, Passive Partial Ownership, and Merger Control, July 2017.

- 259 Link, Thomas and Neyer, Ulrike, Friction-Induced Interbank Rate Volatility under Alternative Interest Corridor Systems, July 2017.
- 258 Diermeier, Matthias, Goecke, Henry, Niehues, Judith and Thomas, Tobias, Impact of Inequality-Related Media Coverage on the Concerns of the Citizens, July 2017.
- 257 Stiebale, Joel and Wößner, Nicole, M&As, Investment and Financing Constraints, July 2017.
- 256 Wellmann, Nicolas, OTT-Messaging and Mobile Telecommunication: A Joint Market? An Empirical Approach, July 2017.
- 255 Ciani, Andrea and Imbruno, Michele, Microeconomic Mechanisms Behind Export Spillovers from FDI: Evidence from Bulgaria, June 2017. Forthcoming in: Review of World Economics.
- 254 Hunold, Matthias and Muthers, Johannes, Capacity Constraints, Price Discrimination, Inefficient Competition and Subcontracting, June 2017.
- 253 Dertwinkel-Kalt, Markus and Köster, Mats, Salient Compromises in the Newsvendor Game, June 2017.
 Published in: Journal of Economic Behavior and Organization, 141 (2017), pp. 301-315.
- 252 Siekmann, Manuel, Characteristics, Causes, and Price Effects: Empirical Evidence of Intraday Edgeworth Cycles, May, 2017.
- 251 Benndorf, Volker, Moellers, Claudia and Normann, Hans-Theo, Experienced vs. Inexperienced Participants in the Lab: Do they Behave Differently?, May 2017. Forthcoming in: Journal of the Economic Science Association.
- 250 Hunold, Matthias, Backward Ownership, Uniform Pricing and Entry Deterrence, May 2017.
- 249 Kreickemeier, Udo and Wrona, Jens, Industrialisation and the Big Push in a Global Economy, May 2017.
- 248 Dertwinkel-Kalt, Markus and Köster, Mats, Local Thinking and Skewness Preferences, April 2017.
- 247 Shekhar, Shiva, Homing Choice and Platform Pricing Strategy, March 2017.
- 246 Manasakis, Constantine, Mitrokostas, Evangelos and Petrakis, Emmanuel, Strategic Corporate Social Responsibility by a Multinational Firm, March 2017.
- 245 Ciani, Andrea, Income Inequality and the Quality of Imports, March 2017.
- 244 Bonnet, Céline and Schain, Jan Philip, An Empirical Analysis of Mergers: Efficiency Gains and Impact on Consumer Prices, February 2017.
- Benndorf, Volker and Martinez-Martinez, Ismael, Perturbed Best Response Dynamics in a Hawk-Dove Game, January 2017.
 Published in: Economics Letters, 153 (2017), pp. 61-64.
- 242 Dauth, Wolfgang, Findeisen, Sebastian and Suedekum, Jens, Trade and Manufacturing Jobs in Germany, January 2017. Forthcoming in: American Economic Review, Papers & Proceedings.
- 241 Borrs, Linda and Knauth, Florian, The Impact of Trade and Technology on Wage Components, December 2016.

- 240 Haucap, Justus, Heimeshoff, Ulrich and Siekmann, Manuel, Selling Gasoline as a By-Product: The Impact of Market Structure on Local Prices, December 2016.
- 239 Herr, Annika and Normann, Hans-Theo, How Much Priority Bonus Should be Given to Registered Organ Donors? An Experimental Analysis, November 2016.
- 238 Steffen, Nico, Optimal Tariffs and Firm Technology Choice: An Environmental Approach, November 2016.
- 237 Behrens, Kristian, Mion, Giordano, Murata, Yasusada and Suedekum, Jens, Distorted Monopolistic Competition, November 2016.
- 236 Beckmann, Klaus, Dewenter, Ralf and Thomas, Tobias, Can News Draw Blood? The Impact of Media Coverage on the Number and Severity of Terror Attacks, November 2016. Forthcoming in: Peace Economics, Peace Science and Public Policy.
- 235 Dewenter, Ralf, Dulleck, Uwe and Thomas, Tobias, Does the 4th Estate Deliver? Towars a More Direct Measure of Political Media Bias, November 2016.
- 234 Egger, Hartmut, Kreickemeier, Udo, Moser, Christoph and Wrona, Jens, Offshoring and Job Polarisation Between Firms, November 2016.
- 233 Moellers, Claudia, Stühmeier, Torben and Wenzel, Tobias, Search Costs in Concentrated Markets An Experimental Analysis, October 2016.
- 232 Moellers, Claudia, Reputation and Foreclosure with Vertical Integration Experimental Evidence, October 2016.
- 231 Alipranti, Maria, Mitrokostas, Evangelos and Petrakis, Emmanuel, Non-comparative and Comparative Advertising in Oligopolistic Markets, October 2016. Forthcoming in: The Manchester School.
- 230 Jeitschko, Thomas D., Liu, Ting and Wang, Tao, Information Acquisition, Signaling and Learning in Duopoly, October 2016.
- 229 Stiebale, Joel and Vencappa, Dev, Acquisitions, Markups, Efficiency, and Product Quality: Evidende from India, October 2016.
- 228 Dewenter, Ralf and Heimeshoff, Ulrich, Predicting Advertising Volumes: A Structural Time Series Approach, October 2016. Published in: Economics Bulletin, 37 (2017), Volume 3.
- 227 Wagner, Valentin, Seeking Risk or Answering Smart? Framing in Elementary Schools, October 2016.
- 226 Moellers, Claudia, Normann, Hans-Theo and Snyder, Christopher M., Communication in Vertical Markets: Experimental Evidence, July 2016. Published in: International Journal of Industrial Organization, 50 (2017), pp. 214-258.
- 225 Argentesi, Elena, Buccirossi, Paolo, Cervone, Roberto, Duso, Tomaso and Marrazzo, Alessia, The Effect of Retail Mergers on Prices and Variety: An Ex-post Evaluation, June 2016.
- 224 Aghadadashli, Hamid, Dertwinkel-Kalt, Markus and Wey, Christian, The Nash Bargaining Solution in Vertical Relations With Linear Input Prices, June 2016. Published in: Economics Letters, 145 (2016), pp. 291-294.

- 223 Fan, Ying, Kühn, Kai-Uwe and Lafontaine, Francine, Financial Constraints and Moral Hazard: The Case of Franchising, June 2016. Forthcoming in: Journal of Political Economy.
- 222 Benndorf, Volker, Martinez-Martinez, Ismael and Normann, Hans-Theo, Equilibrium Selection with Coupled Populations in Hawk-Dove Games: Theory and Experiment in Continuous Time, June 2016. Published in: Journal of Economic Theory, 165 (2016), pp. 472-486.
- 221 Lange, Mirjam R. J. and Saric, Amela, Substitution between Fixed, Mobile, and Voice over IP Telephony – Evidence from the European Union, May 2016. Published in: Telecommunications Policy, 40 (2016), pp. 1007-1019.
- 220 Dewenter, Ralf, Heimeshoff, Ulrich and Lüth, Hendrik, The Impact of the Market Transparency Unit for Fuels on Gasoline Prices in Germany, May 2016. Published in: Applied Economics Letters, 24 (2017), pp. 302-305.
- 219 Schain, Jan Philip and Stiebale, Joel, Innovation, Institutional Ownership, and Financial Constraints, April 2016.
- 218 Haucap, Justus and Stiebale, Joel, How Mergers Affect Innovation: Theory and Evidence from the Pharmaceutical Industry, April 2016.
- 217 Dertwinkel-Kalt, Markus and Wey, Christian, Evidence Production in Merger Control: The Role of Remedies, March 2016.
- 216 Dertwinkel-Kalt, Markus, Köhler, Katrin, Lange, Mirjam R. J. and Wenzel, Tobias, Demand Shifts Due to Salience Effects: Experimental Evidence, March 2016. Published in: Journal of the European Economic Association, 15 (2017), pp. 626-653.
- 215 Dewenter, Ralf, Heimeshoff, Ulrich and Thomas, Tobias, Media Coverage and Car Manufacturers' Sales, March 2016. Published in: Economics Bulletin, 36 (2016), pp. 976-982.
- 214 Dertwinkel-Kalt, Markus and Riener, Gerhard, A First Test of Focusing Theory, February 2016.
- 213 Heinz, Matthias, Normann, Hans-Theo and Rau, Holger A., How Competitiveness May Cause a Gender Wage Gap: Experimental Evidence, February 2016. Forthcoming in: European Economic Review, 90 (2016), pp. 336-349.
- 212 Fudickar, Roman, Hottenrott, Hanna and Lawson, Cornelia, What's the Price of Consulting? Effects of Public and Private Sector Consulting on Academic Research, February 2016.
- 211 Stühmeier, Torben, Competition and Corporate Control in Partial Ownership Acquisitions, February 2016. Published in: Journal of Industry, Competition and Trade, 16 (2016), pp. 297-308.
- 210 Muck, Johannes, Tariff-Mediated Network Effects with Incompletely Informed Consumers, January 2016.
- Dertwinkel-Kalt, Markus and Wey, Christian, Structural Remedies as a Signalling Device, January 2016.
 Published in: Information Economics and Policy, 35 (2016), pp. 1-6.
- 208 Herr, Annika and Hottenrott, Hanna, Higher Prices, Higher Quality? Evidence From German Nursing Homes, January 2016. Published in: Health Policy, 120 (2016), pp. 179-189.

- 207 Gaudin, Germain and Mantzari, Despoina, Margin Squeeze: An Above-Cost Predatory Pricing Approach, January 2016. Published in: Journal of Competition Law & Economics, 12 (2016), pp. 151-179.
- 206 Hottenrott, Hanna, Rexhäuser, Sascha and Veugelers, Reinhilde, Organisational Change and the Productivity Effects of Green Technology Adoption, January 2016. Published in: Energy and Ressource Economics, 43 (2016), pp. 172–194.
- 205 Dauth, Wolfgang, Findeisen, Sebastian and Suedekum, Jens, Adjusting to Globalization – Evidence from Worker-Establishment Matches in Germany, January 2016.
- 204 Banerjee, Debosree, Ibañez, Marcela, Riener, Gerhard and Wollni, Meike, Volunteering to Take on Power: Experimental Evidence from Matrilineal and Patriarchal Societies in India, November 2015.
- 203 Wagner, Valentin and Riener, Gerhard, Peers or Parents? On Non-Monetary Incentives in Schools, November 2015.
- 202 Gaudin, Germain, Pass-Through, Vertical Contracts, and Bargains, November 2015. Published in: Economics Letters, 139 (2016), pp. 1-4.
- 201 Demeulemeester, Sarah and Hottenrott, Hanna, R&D Subsidies and Firms' Cost of Debt, November 2015.
- 200 Kreickemeier, Udo and Wrona, Jens, Two-Way Migration Between Similar Countries, October 2015. Forthcoming in: World Economy.
- Haucap, Justus and Stühmeier, Torben, Competition and Antitrust in Internet Markets, October 2015.
 Published in: Bauer, J. and M. Latzer (Eds.), Handbook on the Economics of the Internet, Edward Elgar: Cheltenham 2016, pp. 183-210.
- 198 Alipranti, Maria, Milliou, Chrysovalantou and Petrakis, Emmanuel, On Vertical Relations and the Timing of Technology, October 2015. Published in: Journal of Economic Behavior and Organization, 120 (2015), pp. 117-129.
- 197 Kellner, Christian, Reinstein, David and Riener, Gerhard, Stochastic Income and Conditional Generosity, October 2015.
- 196 Chlaß, Nadine and Riener, Gerhard, Lying, Spying, Sabotaging: Procedures and Consequences, September 2015.
- 195 Gaudin, Germain, Vertical Bargaining and Retail Competition: What Drives Countervailing Power?, May 2017 (First Version September 2015). Forthcoming in: The Economic Journal.
- 194 Baumann, Florian and Friehe, Tim, Learning-by-Doing in Torts: Liability and Information About Accident Technology, September 2015.
- 193 Defever, Fabrice, Fischer, Christian and Suedekum, Jens, Relational Contracts and Supplier Turnover in the Global Economy, August 2015. Published in: Journal of International Economics, 103 (2016), pp. 147-165.
- 192 Gu, Yiquan and Wenzel, Tobias, Putting on a Tight Leash and Levelling Playing Field: An Experiment in Strategic Obfuscation and Consumer Protection, July 2015. Published in: International Journal of Industrial Organization, 42 (2015), pp. 120-128.
- 191 Ciani, Andrea and Bartoli, Francesca, Export Quality Upgrading under Credit Constraints, July 2015.

- 190 Hasnas, Irina and Wey, Christian, Full Versus Partial Collusion among Brands and Private Label Producers, July 2015.
- Dertwinkel-Kalt, Markus and Köster, Mats, Violations of First-Order Stochastic Dominance as Salience Effects, June 2015.
 Published in: Journal of Behavioral and Experimental Economics, 59 (2015), pp. 42-46.
- 188 Kholodilin, Konstantin, Kolmer, Christian, Thomas, Tobias and Ulbricht, Dirk, Asymmetric Perceptions of the Economy: Media, Firms, Consumers, and Experts, June 2015.
- 187 Dertwinkel-Kalt, Markus and Wey, Christian, Merger Remedies in Oligopoly under a Consumer Welfare Standard, June 2015.
 Published in: Journal of Law, Economics, & Organization, 32 (2016), pp. 150-179.
- 186 Dertwinkel-Kalt, Markus, Salience and Health Campaigns, May 2015. Published in: Forum for Health Economics & Policy, 19 (2016), pp. 1-22.
- 185 Wrona, Jens, Border Effects without Borders: What Divides Japan's Internal Trade? May 2015.
- Amess, Kevin, Stiebale, Joel and Wright, Mike, The Impact of Private Equity on Firms' Innovation Activity, April 2015.
 Published in: European Economic Review, 86 (2016), pp. 147-160.
- 183 Ibañez, Marcela, Rai, Ashok and Riener, Gerhard, Sorting Through Affirmative Action: Three Field Experiments in Colombia, April 2015.
- 182 Baumann, Florian, Friehe, Tim and Rasch, Alexander, The Influence of Product Liability on Vertical Product Differentiation, April 2015. Published in: Economics Letters, 147 (2016), pp. 55-58 under the title "Why Product Liability May Lower Product Safety".
- 181 Baumann, Florian and Friehe, Tim, Proof beyond a Reasonable Doubt: Laboratory Evidence, March 2015.
- 180 Rasch, Alexander and Waibel, Christian, What Drives Fraud in a Credence Goods Market? Evidence from a Field Study, March 2015.
- 179 Jeitschko, Thomas D., Incongruities of Real and Intellectual Property: Economic Concerns in Patent Policy and Practice, February 2015. Forthcoming in: Michigan State Law Review.
- 178 Buchwald, Achim and Hottenrott, Hanna, Women on the Board and Executive Duration – Evidence for European Listed Firms, February 2015.
- 177 Heblich, Stephan, Lameli, Alfred and Riener, Gerhard, Regional Accents on Individual Economic Behavior: A Lab Experiment on Linguistic Performance, Cognitive Ratings and Economic Decisions, February 2015. Published in: PLoS ONE, 10 (2015), e0113475.
- 176 Herr, Annika, Nguyen, Thu-Van and Schmitz, Hendrik, Does Quality Disclosure Improve Quality? Responses to the Introduction of Nursing Home Report Cards in Germany, February 2015. Published in: Health Policy, 120 (2016), pp.1162-1170.
- Herr, Annika and Normann, Hans-Theo, Organ Donation in the Lab: Preferences and Votes on the Priority Rule, February 2015.
 Published in: Journal of Economic Behavior and Organization, 131 Part B (2016), pp. 139-149.

- 174 Buchwald, Achim, Competition, Outside Directors and Executive Turnover: Implications for Corporate Governance in the EU, February 2015.
- 173 Buchwald, Achim and Thorwarth, Susanne, Outside Directors on the Board, Competition and Innovation, February 2015.
- 172 Dewenter, Ralf and Giessing, Leonie, The Effects of Elite Sports Participation on Later Job Success, February 2015.
- 171 Haucap, Justus, Heimeshoff, Ulrich and Siekmann, Manuel, Price Dispersion and Station Heterogeneity on German Retail Gasoline Markets, January 2015. Forthcoming in: The Energy Journal.
- 170 Schweinberger, Albert G. and Suedekum, Jens, De-Industrialisation and Entrepreneurship under Monopolistic Competition, January 2015. Published in: Oxford Economic Papers, 67 (2015), pp. 1174-1185.

Older discussion papers can be found online at: <u>http://ideas.repec.org/s/zbw/dicedp.html</u>

Heinrich-Heine-University of Düsseldorf

Department of Economics

Universitätsstraße 1_ 40225 Düsseldorf www.dice.hhu.de